

# RIETI Discussion Paper Series 22-E-017

# Achieving Inclusive Transportation: Fully Automated Vehicles with Social Support

**YOO, Sunbin** Urban Institute, Kyushu University

**KUMAGAI, Junya** Urban Institute, Kyushu University

**KAWABATA, Yuta** Urban Institute, Kyushu University

> MANAGI, Shunsuke RIETI



The Research Institute of Economy, Trade and Industry https://www.rieti.go.jp/en/

#### Achieving Inclusive Transportation: Fully Automated Vehicles with Social Support\*

Sunbin YOO<sup>1</sup>, Junya KUMAGAI<sup>1</sup>, Yuta KAWABATA<sup>1</sup> and Shunsuke MANAGI<sup>1.2</sup> <sup>1</sup> Urban Institute, Kyushu University <sup>2</sup> Research Institute of Economy, Trade and Industry

#### Abstract

We provide quantitative evidence of whether a representative sample of the newly introduced fully automatic vehicles (FAVs) are inclusive. We answer this question by examining FAV demand with a focus on natural disaster victims—people who have become physically or mentally challenged due to severe disaster damage, including those with post-traumatic stress disorder. We investigate whether the fear of natural disasters, social support, environmental concerns, the fear of potential accidents, and merits regarding FAVs are motivators of, or hindrances to, purchasing intentions of FAVs. To do so, we acquire a unique dataset covering disaster victims with traumatic disaster damages (12,286 observations in total) and people without such experiences (57,105 observations in total). Then, we construct a multigroup structural estimation model to estimate FAV demand. We conduct estimations of latent and socioeconomic variables which demonstrate people's attitudes. Our findings show that the social support of family, friends, and local authorities is a crucial factor in motivating disaster victims to appreciate and purchase FAVs. The positive impact of social support on appreciating/purchasing FAVs can offset the negative impacts of a fear of natural disasters and accidents, thus enabling more people to enjoy FAVs.

Keywords: Autonomous vehicle; Automated transport; Self-driving car; Emerging transport modes; Mobility on-demand; Demand Estimation JEL classification: L62, Q48, Q55, Q58

The RIETI Discussion Paper Series aims at widely disseminating research results in the form of professional papers, with the goal of stimulating lively discussion. The views expressed in the papers are solely those of the author(s), and neither represent those of the organization(s) to which the author(s) belong(s) nor the Research Institute of Economy, Trade and Industry.

<sup>\*</sup>This study is conducted as a part of the Project "Institutional design for desirable acceptance of AI technology" undertaken at the Research Institute of Economy, Trade and Industry (RIETI). The authors are grateful for helpful comments and suggestions by Discussion Paper seminar participants at RIETI.

# **1** Introduction

#### **1.1 Research Motivation**

The introduction of new transportation modes demands *inclusiveness*, which indicates the inclusion of the socially vulnerable ([90]). Excluding the vulnerable would result in transport disadvantages, transport poverty, and an increase in the risk of social exclusion, widening the disparities between those who belong to the vulnerable and the general public.

However, despite the soaring attention paid to the potential benefits, whether such modes are inclusive remains unanswered. Previous works express concern on the transportation exclusion problems faced by low-income individuals, which is a crucial problem also addressed in [53], [77]. However, the discussions on the social exclusion of those who are not allowed to drive, regardless of their sociodemographic status, due to their physical or cognitive disabilities remain inadequate. Research gaps persist in whether these people can fully enjoy the newly introduced (or planned to be introduced) modes. Among several newly introduced transportation modes, this study focuses on fully-automated vehicle (FAV) technology – which do not require human operators and thus can serve the vulnerable – which has the potential to fundamentally alter transportation systems by reducing fatal road accidents, provide critical mobility to elderly and disabled people, increase road capacity, save fuel, and reduce emissions ([4]).

#### **1.2 Previous Works and Our Contribution**

The aforementioned potential benefits have allowed FAVs to consistently capture the attention of the public and researchers. Recent studies identify ways to extend the market share of FAVs and autonomous vehicles (AVs) by understanding consumer choices ([97], [29], [8]), investigating the diverse perceptions of F/AVs ([19], [13], [15], [69], [32]), and identifying the demand for various types of AVs such as shared AVs (SAVs)([85], [22], [47]). Most recent strands of literature argue that a high level of environmental consciousness, a high education level, younger age, and a higher level of income are positively correlated with consumers' willingness-to-buy (WTB) and willingness-to-pay (WTP) of F/AVs.

Thus far, few studies have investigated whether socially vulnerable classes would adopt and appreciate FAVs. A notable exception is [17], which focuses on the role of AVs for elders in overcoming hindrances to travel. This is because older adults are likely to experience transport disadvantages, transport poverty, and the risk of social exclusion because their physical or cognitive disabilities might prevent them from using FAVs ([76]). While such findings are worthwhile, in this study, we focus on the aspects of social exclusion. Specifically, what is the likely outcome for individuals experiencing physical or cognitive disability, regardless of age, income level, gender, and other sociodemographic characteristics? Simply put, people with cognitive impairments might find it difficult to operate new technologies and to override controls ([64]).

In this study, we focus on *disaster victims* who have experienced traumatic damages due to the natural disasters, which include but are not limited to: the loss of family and friends, severe injuries, and post-traumatic stress disorder (PTSD) and are thus likely to have temporarily or permanently lost the ability to drive. Would factors closely correlated with the public acceptance of F/AVs still be crucial for the disaster victims to accept/appreciate F/AVs? By answering this question, this study contributes to both understanding demand patterns and the provision of policies designed to encourage inclusive transportation modes, which goes beyond discussions on FAVs.

Additionally, we follow numerous previous studies that report that natural disaster experiences increase risk perceptions ([27], [23], [9], [33]) and that increased risk perceptions discourage victims from accepting new technology ([50], [11]). Such trends can also be found in the field of transportation (which is usually represented by a new mode choice or driving behavior) ([57], [58], [2]) Among these studies, previous works on FAVs report that increased risk would hinder the use and adoption of FAVs ([14], [94], [93]).

#### **1.3 Research Hypothesis**

Referring to the findings of and gaps in previous works, we set our main research questions as follows: If disaster victims are reluctant to employ FAVs, then would support from family, friends, local authorities, and the government encourage them to choose or appreciate FAVs? Previous works argue that fears about new technology among disaster victims would hinge substantially on social support, namely support from family, friends and local authorities (governments). Thus, social support may decrease perceived risks ([86], [24], [66]). If that is the case, whether social support can encourage the disaster victims to adopt new technologies is also worth investigating.

Additionally, we examine whether disaster victims are reluctant to choose FAVs because FAVs may take control of the drivers in times of an accident. The previous works often witness such trends, which mention that the disaster victims are substantially concerned about controllability in protecting themselves ([27], [38]).

#### **1.4 Brief Introduction on Methodology**

We conduct a large-scale survey of over 100,000 respondents in Japan. We first ask questions on disaster experiences, such as whether a respondent is a disaster victim and as a result has experienced PTSD, physical injuries, mental damages, severe losses of private property and the loss of friends/family members. Because we select Japan as our study area, which is particularly vulnerable to natural disasters because of its climate and topography with experiences of countless earthquakes, typhoons, and other types of disasters, our data allow us to obtain a sample that includes over 10,000 observations of disaster victims. This setting allows our results to be more realistic because losing the ability to drive due to a disaster may require multiple natural disaster experiences, which are likely to happen in Japan. Thus, our setting eliminates the necessity of soliciting the time and name of a specific disaster that caused physical and mental damages to the respondent. Next, we additionally ask questions regarding the purchasing decisions (willingness-tobuy, WTB) and perceived value of FAVs (hereafter, FAV values), individual characteristics such as income, gender, age, environmental concerns, and opinions on the advantages of and concerns regarding FAVs.

Referring to the extensive literature reviews on the factors that are crucial for FAV demand, we construct four empirical models that answer our research questions. After filtering out the respondents who were unaware of FAVs, lacked environmental concerns or failed to provide sociodemographic details, we build models for full samples (69,391 observations) and three sample groups: those without disaster experiences (Group 1, 57,105 observations), disaster victims, who are classified according to the type of disaster experience (Group 2, 7,853 observations, and Group 3, 4,433 observations) to compare coefficients across the sample groups. We graphically illustrate our research in Figure 1. We first identify attitudes with respect to behaviors and estimate the relationship between attitudes (along with socioeconomic variables) and decisions.

Intention (Latent): People's Attitudes	Final Decision	Socio-Demographics
Social Supports	Purchasing Decision	Income
Natural Disaster	Entire Sample Group 1: without Disaster Experience	Gender
Traffic Accidents	(57,105 samples) Group 2: Disaster Victims (1) (7,853 samples) Group 3: Disaster Victims (2) (4,433 samples)	Family Information
Merits		Commuting Traits
Natural Conservation	Indicators Specific Behaviors Inferred by Attitudes	Car Ownership
Pollution Alleviation	Indicator survey items (i.e., How much do you think you are risk-	The Number of Children
	averse?, How much do you fear natural disasters?)	Age

Figure 1: Study Structure

#### 1.5 Paper organization

We further discuss how to encourage disaster victims to gradually accept and appreciate FAVs. The remainder of this paper is structured as follows. Section 2 provides background information on the industry and policy. The data and model are presented in Sections 3 and 4. Section 5 reports the empirical results. Section 6 discusses our findings and provides policy implications. Section 7 concludes the paper.

# 2 Backgrounds

In this section, we first explain what an FAV is, then briefly introduce the governmental efforts of Japan in achieving inclusive transport by focusing on FAVs.

#### 2.1 Fully Autonomous Vehicles (FAVs)

AVs have different levels of automatic operation as presented in Figure 2. A system generally unconditionally performs all driving operations at level 5 (fully automated, or FAV) and has not yet been realized as of 2021. Currently, consumers cannot select the specific technology or devices to integrate with their new cars when purchasing level 1 and 2 AVs.<sup>1</sup> For high and full levels of automation (levels 4 and 5), cars are expected to pass driving qualification tests, which inspect whether obligatory automation-related equipment and accessories are equipped. As illustrated in Figure 2, the Society of Automotive Engineers (SAE) has defined different levels of automated functionality, ranging from no automated features (level 0) to full automation (level 5 — commonly referred to as autonomous, self–driving or driverless vehicles).

#### 2.2 FAVs and Social Exclusion

The Japanese government announced that FAVs can create mobility opportunities to socially excluded people who are unable to drive, such as the elderly (those with noticeable

<sup>&</sup>lt;sup>1</sup>We refer to the year 2021.



Figure 2: Summarized Explanations of AV Technology by Level. *Source*: Society of Automotive Engineers International (2021, Accessed August 4, 2021)

problems in driving capacities) and physically/mentally challenged people. As mentioned in Section 1.2, such traits are evident among the disaster victims who are afraid of losing *control*, the ability to protect themselves, while they are inside the FAVs. To do so, the Japanese Government–along with FAV manufacturers– needs to show *social assurance* that FAVs are safe ([56], [91]) in order to accommodate for disaster victims. Such efforts by the Japanese Government to pursue social assurance of FAVs, and their adoption by socially-excluded people, motivate our study to focus on Japan, with the disasterrelated characteristics mentioned in Section 1.2. Meanwhile, because such efforts are increasing in other countries, our implications can be extended to similar production and legislative efforts in other countries trying to pursue FAVs.

The efforts of the Japanese government can be classified into two types; encouraging technological developments and the inclusion of socially vulnerables in the legislation scheme. First, to motivate technological developments that improve the vulnerable in

using FAVs, the Japanese Government has emphasized the necessity of developing technologies capable of resolving social problems by listening to the voices of various stakeholders, including those *who find it challenging to speak up socially*. Through cabinet meetings and policy guidelines, the Japanese Government has urged automotive manufacturers to develop technologies that can assist socially vulnerable people. An example includes developing systems ensuring the safety of FAVs in times of accidents to the vulnerable such as the elderly, disabled, pregnant women, and so on.<sup>2</sup>

Second, the Japanese Government announced that it would actively promote FAVs as the representative inclusive transportation mode, encompassing those who are socially excluded. On March 29, 2019, the Japanese cabinet announced that the country's entire transportation system should pursue a transition to be more flexible to change and respond as FAV technology evolves, thereby fully realizing the benefits of FAVs. The transition should not only cover improving efficiency based on existing socially accepted objectives (e.g., improving convenience and preventing accidents) but also realizing new value created by diversifying the objectives themselves (by including socially vulnerable people in the policy goals) and resolving problems such as transport disparity and social exclusion.

## **3** Data

As mentioned in Section 1, disaster victims may refuse to adopt FAVs regardless of social support due to a fear of new technology adoption, and uncertainty of accidents arising from a loss of direct control of the vehicle. If such concerns outweigh the reduction in risk due to social support, disaster victims would not choose or appreciate FAVs. In this case, a valid empirical strategy is to compare the levels of the psychometric variables while allowing correlations between them; we seek to design empirical strategy successfully

<sup>&</sup>lt;sup>2</sup>We refer to the national report published by the Japanese Government Cabinet Secretariat, "Social Principles of Human-Centric AI," and the public report entitled "SIP Automated Driving for Universal Services (SIP-adus) R&D Plan" published by New Energy and Industry Technology Development Organization (NEDO). Both documents were accessed on August 17, 2021.

does both.

We first conducted an online survey in Japan from November 24 to December 5, 2017. The survey was administered to individuals aged 18 and older in Japan. We randomly selected respondents while maintaining gender and age distributions of the respondents similar to those of the Japanese population. <sup>3</sup> Consequently, we had 100,810 respondents. To assess the sample representativeness of the survey, we present the distribution of socioeconomic variables of the survey and Japanese Census data in Appendix Table A1. We find that there are slight differences in gender and education levels between our survey and Japanese Census data. Before the large-scale survey started, a pre-survey was administered to calibrate the questionnaires. In what follows, we explain our data by dividing the explanation into the areas of 1) identifying disaster victims, 2) FAV-related questions, 3) socioeconomic variables and 4) latent variables.

#### 3.1 Identifying Disaster Victims

To proceed, we first ask whether our respondents had experienced physical/mental damages due to natural disasters. If so, we then asked the respondents to look into a list of disaster damages, allowing the respondents to choose any item that they have experienced, as illustrated in Table 1, which shows the number and the portion of observations associated with each item in the first column. Among the items listed in Table 1, we place particular emphasis on those that are likely to be a 'traumatic experience' which covers experiences are likely to be closely related to physical and mental problems such as anxiety, major depression, nightmares, hypervigilance, and panic attacks associated with trauma-related stimuli, according to previous works in the field of medicine ([46], [61]).

Respondents were allowed to check multiple items. For example, a respondent can check both 'Severe Health Problems' and 'Severe Injury'. Summing up the findings from the previous works aforementioned, we denote the following as 'traumatic experiences':

<sup>&</sup>lt;sup>3</sup>Such a process was possible as we employ an internet survey conducted by Nikkei Research Inc., the largest research company in Japan. Several trap questions were included in the survey to identify respondents who did not seriously answer the questions. Those who did not correctly answer such trap questions were excluded from the survey company's sample in the collection process.

Entire Collapse of a Home (Item No. 1), Home Destruction (Item No. 2), Severe Damage to Property (Item No. 3), Severe Damage to Furniture (Item No. 4.), Death of Family or Friends (Item No. 5), Severe Injury (Item No. 6), Post-traumatic Stress Disorder, PTSD (Item No. 10), and Severe Health Problems (Item No. 11). We place the respondents with these items checked in Group 2. Following previous works, for Group 3, we exclude those who checked destruction of family property (Item No. 3) and severe damage to home furniture (Item No. 4.) from Group 2 to create a category with narrower coverage that includes the items that are more closely associated with PTSD ([75], [25]) that hinder driving ability. Consequently, we have 21,915 and 8,067 respondents in Groups 2 and 3, respectively, which are sufficient numbers of observations for us to conduct statistical analysis, as presented in Table 1.<sup>4</sup> The respondents belonging to each group are mutually exclusive, meaning that no respondent belongs to multiple groups. <sup>5</sup> The difference between Groups 2 and 3 is whether a person experiences partial damage to family property or damage to furniture, which are relatively minor compared to other types of items such as PTSD, the death of a family member, and severe injury.<sup>6</sup> Figure 3 clarifies for readers how we create the groups and which items denote membership in each group, with the proportion of each item after excluding those who had not experienced disaster damage and those who answered: "I don't know." The general pattern of this figure reveals that most damages would belong to Group 2, while Group 3 would cover a narrower range of damages.

<sup>&</sup>lt;sup>4</sup>Note that the total numbers of observations are not equivalent to the 'analytical sample'; that is, we filter out some of the respondents according to certain criteria (i.e., we drop some observations if they have no awareness of FAVs). We further discuss this issue in Section 3.4.

<sup>&</sup>lt;sup>5</sup>Not every person who has experienced a disaster loses driving ability, which would require a substantial level of physical and/or mental damage. Therefore, investigating disaster victims who lost their ability to drive requires dividing the observations into the disaster victims with and without a severe level of disaster experience.

<sup>&</sup>lt;sup>6</sup>We do not address whether Group 2 or 3 has a more severe level of disaster damage. Instead, such differences are likely to be in the type of disaster damage. As mentioned in Section 1, for some people, the loss of driving ability may require a respondent to experience multiple instances of disaster damage, and this is likely to arise in Japan, where natural disasters are common. Therefore, we choose not to ask the respondents for the time and name of a specific disaster that caused their physical and mental damages. More attention should be devoted the which to which a respondent belongs.

No.	Item	Obs.	Portion(%)	Public	Disaster V	/ictims
				Group 1	Group 2	Group 3
1	Entire Home Collapse	1,474	1.47%	Х	0	0
2	House Destruction	2,000	2.00%	Х	0	0
3	Severe Damage to a Property	7,045	7.04%	Х	0	Х
4	Severe Damage to the Furniture	6,921	6.91%	Х	Ο	Х
5	Death of Family and Friends	1,137	1.14%	Х	0	0
6	Severe Injury	1,019	1.02%	Х	0	0
7	Evacuation	1,741	1.74%	Х	Х	Х
8	Moving House	2,429	2.43%	Х	Х	Х
9	Unemployment	1,369	1.37%	Х	Х	Х
10	PTSD (Post-traumatic Stress Disorder)	879	0.88%	Х	0	0
11	Severe Health Problems	836	0.84%	Х	0	0
12	Others	3,591	3.59%	Х	Х	Х
13	No experience.	69,647	70.92%	Ο	Х	Х
14	I don't know.	9,049	7.23%	Х	Х	X
	Total	100,810	100%	69,647	21,915	8,067
	Analytical Sample Total			57,105	7,853	4,433
PTS	D (Post-traumatic Stress Disorder) 2.81%	, 0				
		0				
	Death of a Death New York and 2 51	70  0/				
	Death of a Family Member 3.5	1%				
	An Entire Collapse of a Home 3.8	8%				
	Unemployment 4	.38%				
	Evacuation	5.96%				
	Partial Destruction of a Home	6.33%				
	Moving House	8.10	%			
	Others		10.68%			
	Damage to the Furniture				24.00	)%
Pa	rtial Damage to a Family Property				24.2	3%
	Items belong to neither Group 2 or 3	ms belong t	o Group 2 and 3	Items	belong to G	coup 2

#### Table 1: Criteria for Selecting Disaster Victims

Figure 3: Distribution of Disaster Criteria Items

### 3.2 FAV-related Questions

For the questions related to FAV purchasing intention, respondents were asked the following question: "Do you want to add a completely self-driving option that allows you to move around when you purchase a car in the future?". The respondents were then offered the following response options: "(1) Purchase for sure, (2) Purchase under certain conditions, (3) Do not purchase, and (4) I don't know.". Given that FAVs had not yet been fully introduced into the market in 2017, we assume that people who show an affinity for FAVs can be potential consumers in the future.<sup>7</sup> Therefore, we include those who responded (1) or (2) in a group of 'potential consumers' because they show affinity toward using FAVs. On the other hand, people who responded (3) or(4) are reluctant to purchase FAVs, and we do not consider them potential consumers. Therefore, we code WTB equal to 1 if a respondent belongs to the potential consumer group and 0 otherwise. Using binary outcome variables for SEM is commonly used by previous works in the fields of transportation, as in [60], [59] and [31].

Therefore, our analysis allows us to identify which types of factors would shift consumers who belong to (3) or (4) to (1) or (2). Note that we make a clear distinction between "adding" a completely self-driving option and "purchasing" an FAV by asking "Do you want to add a completely self-driving option that allows you to move around when you purchase a car in the future?". Such a setting allows us to clearly distinguish between 'purchasing a car' and 'adding a self-driving option', which allows us to exclude potential deviations in the results due to the factors associated with purchasing a car, such as car price and vehicle attributes.<sup>8</sup>

Next, we also elicited the *perceived* FAV value (hereafter, FAV value) for FAVs, which refers to the *perceived* value of FAVs expressed in monetary term. Researches in the field of transportation conclude that the perceived value works as a predictor of purchase in-

<sup>&</sup>lt;sup>7</sup>We would like to note that we do not believe that our data are outdated. Given that FAVs were not completely introduced in the market at the time and have yet to be introduced as of 2021, a substantial change in the results, for example, a change in sign or implications, is unlikely. Therefore, more attention should be given to the signs and relative comparisons of coefficient magnitudes of the latent constructs.

<sup>&</sup>lt;sup>8</sup>We would also like to clarify that we do not employ all the information from the survey; that is, we do not treat survey answers on WTB as 'ordinal' but instead treat it as a categorical variable. For example, while it is possible to investigate the result of 'ordinal' responses on WTB by treating 1 not purchasing, 2 as not sure, 3 as considering purchasing and 4 as purchasing, an increase from 1 to 2 does not necessarily indicate the increase of the probability of a respondent purchasing an F/AV. Similarly, do not investigate the multinomial responses regarding WTB because we believe that each response is independent; therefore, investigating how the decision of 'not purchasing' based on another response, for example, 'not sure,' would not fit our research. Thus, we analyze binary responses because we are interested in whether a respondent would purchase an F/AV.

tentions ([40];[12]) and [10] and [20] suggests the role of perceived value in terms of autonomous vehicles and shuttles. Such analyses are capable because perceived values indicate influences on behavioral intentions, which this study is keen to look at through constructing latent variables. Thus, FAV values in this study work as an index showing the appreciation towards FAVs, rather than costs of automation which would show the negative relationship to purchase intentions. Therefore, it is quite natural that higher FAV value would be positively correlated to the higher chances of purchasing FAVs.

In this study, respondents were asked to freely write down their FAV value regardless of their purchasing decisions, leaving it as an open question. As we excluded the respondents unaware of FAVs, concerns regarding unrealistic answers – such as extremely high FAV values – are less likely to appear.

#### 3.3 Sociodemographic Variables

To control for the respondents' sociodemographic characteristics, we also include sociodemographic variables: income, gender, age, number of family members, commuting time and commuting costs.<sup>9</sup> We also include a car ownership dummy (=1 if the respondent owns a car) and car type dummy (hybrid, plug-in hybrid) for two reasons. First, we want to increase the survey's internal validity; therefore, we would like to control for individuals who do not know the price and maintenance costs of cars. Thus, we included the 'car ownership' variable to control for those who do not own a car and are less likely to be aware of car prices. Second, along with car ownership, we also include car types (hybrid and plug-in hybrids (PHEVs)), because car prices differ according across these car types.<sup>10</sup>

While we only survey those who were aware of FAVs, it is possible that in 2017, the respondents were less familiar with FAVs than their counterparts in 2021 would be. Sorting out respondents who were not familiar with the FAV is necessary. Thus, we exclude

<sup>&</sup>lt;sup>9</sup>For those who do not commute, the commuting time is treated as zero.

<sup>&</sup>lt;sup>10</sup>We do not investigate other car types for two reasons: First, 90% of the Japanese car market consists of gasoline cars. Second, we do not observe substantial differences between gasoline cars and cars with other fuel types, such as diesel.

those who answered 'I don't know' to all of the questions related to FAVs (5,987 observations). Finally, we drop those who selected "don't know/don't want to answer" about their individual income (30,156 observations). As a result, we have 69,391 respondents in total.

Finally, we summarize our data in Table 2, which shows descriptive statistics of the key variables for full sample (the first column) and each group in the following columns. The average FAV values range from around 658,830 JPY to 868,030 JPY. As of October 2020, Tesla priced their full-driving capability (level 4 if all advertised features are delivered) as 871,000 JPY, in Tesla Model 3 (of vehicle price 4,590,000 JPY) <sup>11</sup> While our FAV value is lower than that of Tesla's, there is a fundamental difference between Tesla's case and the survey investigated in this study. Tesla is charging consumers with the promise that level-4 automation will be delivered in the future, and it is indeed based on the trust of Tesla, which is indeed a famous company. Compared to Tesla, respondents in our study would have uncertainty particularly in year 2017; therefore, it is clear that automation technology defined in this study should be appreciated lower. However, because our FAV values do not vary substantially compared to Tesla's market price, we would like to make a clear note that our FAV value can be used for the analysis.

The summary statistics in Table 2 indicate that the average WTB and FAV value of disaster victims (Groups 2 and 3) are higher than those of respondents who are not disaster victims (Group 1). Although the dummy variables are excluded from this table, we include the dummy variables for hybrid cars and PHEVs, a long commute dummy if commuting time exceeds 2 hours, and a short commute dummy if commuting time is less than an hour.

Interestingly, we find that both the WTB and FAV value increase as the disaster damage increases (moving from Group 1 to 2 and Group 1 to 3), indicating that disaster victims recognize the benefits of choosing FAVs. We do not observe substantial differences between WTB across groups; for example, the WTB of Groups 2 (0.533) and 3 (0.547) are

 $<sup>^{11}</sup>$ We refer to the Tesla official website: Tesla, 2020. Autopilot. Retrieved 2020, from  $https://www.tesla.com/en_ca/model3/design#autopilot.$ 

11.7% and 14.47% higher than the Group 1 (0.477), respectively. However, we observe significant differences in FAV value between the full sample and Group 3, as the FAV value of Group 3 (86.803) is approximately 31.75% higher than the full Sample average (65.883). On the other hand, the FAV value of Group 2 does not show substantial differences from the full sample, as the FAV value of Group 2 is 5.23% (69.327) higher than the full sample average. Furthermore, while FAV value in Group 3 seems higher than in the full sample, the standard deviation is also higher (163.358 in Group 3 and 128.386 in Group 1), indicating the FAV value distribution within Group 3 is uneven. What factors motivate disaster victims to adopt and appreciate FAVs? Why are there more negligible differences within sociodemographic variables (e.g., income, age, commuting time,), do psychometric attitudes influence WTB and FAV value? We answer these questions empirically by constructing latent variables and structural equation models explained in the following Subsections 3.4 and 4.1.)

#### 3.4 Constructing Latent Variables

As mentioned in Subsection 3.3, we do not find significant differences in sociodemographic variables across groups. Therefore, we proceed to the empirical analysis of psychometric factors and construct latent variables. We identify the latent variables that can be related to WTB for FAVs based, whenever possible, on statements previously used and found to be effective in the literature(e.g., [51], [45]). Table 3 presents the names (acronyms) of our latent variables, items corresponding to them, and a list of references. In the following two paragraphs, we explain how we identify the latent constructs. The survey consists of questionnaires related to the following:

1. **Natural Disaster**: Whether a respondent has experienced a natural disaster that results in the loss of family, friends, and severe physical damage to the home and assets and whether the respondent has PTSD. It also captures whether and how fearful a respondent is, towards natural disaster.

Variable	Mean	Std. Dev.	Min	Max
Full Samples (n=69,391)				
WTB (=1 if purchase FAVs)	0.488	0.500	0	1
FAV Values (10,000 JPY)	67.618	131.359	0	999
Female (=1 if female)	0.307	0.461	0	1
Annual Income (10,000 JPY)	511.958	420.924	100	3,500
Age	0.486	0.506	0	3.25
Commuting Time (Hours)	50.002	11.264	17	99
Household Members	2.785	1.315	1	10
Household Members: Preschoolers	0.132	0.432	0	5
Commuting Cost (1,000 JPY)	0.620	2.360	0	999
Car Ownership	0.802	0.398	0.00	1
Group 1 (n=57,105)				
WTB (=1 if purchase FAVs)	0.477	0.499	0	1
FAV Values (10,000 JPY)	65.883	128.386	0	999
Female (=1 if female)	0.315	0.465	0	1
Annual Income (10,000 JPY)	511.001	415.594	100	3,500
Age	49.747	11.170	17	99
Commuting Time (Hours)	0.487	0.505	0	3.25
Household Members	2.784	1.313	1	10
Household Members: Preschoolers	0.134	0.433	0	5
Commuting Cost (1,000 JPY)	0.614	2.309	0	99
Car Ownership	0.796	0.403	0	1
Group 2 (n=7,853)	0 500	0.400	0	1
WIB (=1 II purchase FAVS)	0.533	0.499	0	1
FAV values (10,000 JPY)	0.264	131.588	0	999
Appuel Income (10,000 IDV)	0.204	0.441	0	1
Aminual mcome (10,000 JP 1)	514.520 52.450	422.010	100	3,300 06
Age Commuting Time (Hours)	0 450	0.502	17	30
Household Members	0.455	1 300	1	3.23 10
Household Members: Preschoolers	2.015	0.415	0	5
Commuting Cost (1 000 IPV)	0.117	2 239	0	5 80
Car Ownershin	0.331	0.358	0	1
Group 3 (n=4.433)	0.015	0.000	0	1
WTB (=1 if purchase FAVs)	0.547	0.498	0	1
FAV Values (10,000 JPY)	86.803	165.358	0	999
Female (=1 if female)	0.285	0.452	0	1
Annual Income (10,000 JPY)	520.088	481.752	100	3,500
Age	48.958	11.614	17	99
Commuting Time (Hours)	0.515	0.521	0	3.25
Household Members	2.754	1.366	1	10
Household Members: Preschoolers	0.139	0.452	0	5
Commuting Cost (1,000 JPY)	0.736	3.097	0	80
Car Ownership	0.801	0.399	0	1

Table 2: Descriptive Statistics of Analytical Sample

2. **Social Support**: The extent to which a respondent trusts the government, local authorities, family and friends during disasters.

- 3. **FAV-related questions**: Expected benefits of FAVs or merits, and whether the respondents fear FAVs due to potential accidents (Accidents).
- 4. Environmental Concerns:Importance of environmental protection (Nature), importance of reducing environmental pollution (Pollution).

Table 3 presents the list of questions and the previous works to which we referred to create survey questions according to the latent categories. For the (fear of) natural disaster (ND) and social support (SP) variables, respondents were asked to answer about their fear of natural disasters and beliefs in social support from diverse authorities on a scale from 1 to 5, where (1) indicates strongly disagree; (2) indicates disagree; (3) indicates neither agree or disagree; (4) indicates agree; and (5) indicates strongly agree. For example, if a respondent believes that his/her life would be in danger due to a large-scale natural disaster (Question No. 1 in the Natural Disaster Item), s/he would mark (5).

Next, we asked concerns regarding the environment in terms of importance as a policy. Based on previous studies, we classified the environmental policy topics into eight factors referring to the House of Councilors, The National Diet of Japan, (2015). We have 13 questions in total, and the topics cover air pollution, environmental conservation, water pollution, endangered species conservation (biodiversity), reuse and recycling, waste disposal, and  $CO_2$  emissions with questions such as, "How important is the policy to you?" The scale of responses is as follows: (0) for no awareness/interest at all – meaning that the difference between those who answer (0) and others would be whether that person at least has an interest in a certain policy/issue – (1) for very insignificant; (2) for insignificant; (3) for neither important nor insignificant; (4) for important; and (5) for very important. According to the context of the questions, we divide the questions into pollution (EP), if a question is related to air pollution and waste management, and nature (EN) if a survey item is related to biodiversity and natural environment conservation.

#### Table 3: Latent Variables, Explanation and Works Referenced

Explanation	Source
Latents on Risk Perceptions	
<i>Natural Disaster (ND)</i> (Cronbach's $\alpha = 0.6961$ )	[27], [23], [9], [11], [33], [50]
ND1. Do you think your life will be in danger due to a large-scale natural disaster?	
ND2. Do you think your property (household, household goods, automobiles, etc.)	
will be damaged by a large-scale natural disaster?	
ND3. Do you think a large-scale natural disaster will	
isolate you from your surroundings?	
ND4. Do you think you will need to evacuate from your home	
during a disaster to save your life?	
Social Support (SP) ( $\alpha = 0.6579$ )	[3], [5], [28], [39], [78], [83]
When faced with difficulties such as	
supplies, money, and housing in times of disaster,	
choose the one that best describes your thoughts.	
<b>SP1.</b> I expect physical/mental support from the government,	
local authorities and public institutions.	
<b>SP2.</b> I expect physical/mental support from family members and friends.	
<b>SP3.</b> I expect physical/mental support from local volunteers and	
members of local communities.	
Latents on FAVs	
<b><i>Fear (FE)</i></b> ( $\alpha = 0.7345$ )	[44], [34], [95], [49], [84], [43]
FEL. There is a possibility that children will be able to move it on their own.	
FE2. There is a possibility that the software will be hacked (cyber security).	
<b>FE3.</b> A malfunction may cause accidents.	
FE4. It is unclear who is responsible for an accident due to FAV technology.	
<b>FE5.</b> Traffic volume and congestion might increase because those without a license can drive.	
<b>FE6.</b> A malfunction may take me to the wrong destination.	
<i>Merits (MR)</i> ( $\alpha = 0.5658$ )	[96], [52], [87], [81], [41], [68]
MR1. People can drive without a license.	
MR2. Burdens on driving would be decreased.	
MR3. Children can move the vehicle without a guardian.	
<b>MR4.</b> Able to do other work while driving. (Multitask)	
MR5. Able to avoid responsibility for traffic accidents.	
<b>Pollution (EP)</b> ( $\alpha = 0.9558$ )	[92], [36], [36], [42], [74]
EP1. Recycling is important.	
<b>EP2.</b> Cycle utilization rate (the percentage of the total amount of	
reusable and recycled materials to be injected into society),	
is important for preventing pollution.	
<b>EP3.</b> I think water quality should be improved.	
<b>EP4.</b> Alleviating particulate matter (PM 2.5.) pollution is critical for our society.	
EP5. Resolving air pollution (particularly photochemical smog) is important.	
Nature (FN) $(\alpha - 0.9188)$	[67] [71] [80] [18]
<b>FN1</b> Dreserving and angered species is important	[07], [71], [00], [10]
Even reserving by an imple (overall) is important	
<b>EN2.</b> Firstiving number diminals (overlain) is important.	
ENG. The faile of green area within 1,500 meters of a nouse is important.	
Live, orech purchashig, when purchashig goods and services,	

# 4 Empirical Strategy

In this section, we introduce our empirical strategy, then explain how we identify the latent variables in our empirical model.

#### 4.1 Methods

We choose structural equation modeling (SEM) to assess the relationship between factors that are correlated with the WTB and FAV value. SEM allows us to examine the psychometric factors that are correlated with people's intentions regarding FAVs. SEM can handle a substantial number of endogenous and exogenous variables and can include latent variables in the model, and is useful to analyze the relationship between individual's intention and behavior. Thanks to these benefits, SEM has been employed in many research fields incorporating psychometric modeling, such as psychology, sociology, educational research, political science, and marketing research.

Moreover, SEM offers simultaneous estimations of latent variables and exogenous variables and allows for correlations between latents. If the latent and exogenous variables are estimated sequentially, for example, one can conduct factor analysis to construct the latents in the first step and proceed to the estimation of latent and exogenous variables in the choice modeling; while this strategy is simple, it does not guarantee unbiased estimators of the parameters involved and tends to underestimate standard errors (see, for example, [88], [54]). Finally, we use a multigroup SEM analysis to examine the differences between disaster victims and those without disaster experiences. In that sense, we follow the empirical strategies used in the previous works on FAV demand applying SEM ([16], [98], and [55]). We use STATA and a maximum-likelihood model to estimate the model (see [72] for a discussion of sequential versus simultaneous estimation).

Based on the findings from the literature mentioned in Sections 1 and 2, we choose six latent variables: Natural Disaster, Social Support, Nature, Pollution, Merit, and Accidents. Because these items focus on the psychometric intentions of the potential consumers, we choose SEM, which allows such analysis.



Figure 4: Graphically Explained our Brief Model Structures

#### 4.2 Structural Equation Modeling

We used the latent constructs to create SEM models. We have four models in total. Figure 4 presents a brief structure of our SEM models. First, we investigate factors that are correlated with WTB (Model (1)) and FAV value (Model (2)). Second, we assume that FAV value would be correlated with WTB; therefore, we add such a relationship to Model (1) (Model (3)). Finally, we include Model (4), which assumes that all types of latents and other exogenous variables are correlated with both WTB and FAV value on top of Model (3). Our preferred main model is Model (4), and we use Models (1) to (3) to confirm our findings from Model (4). The diverse specifications employed in Models (1) to (3) allow us to confirm the robustness of the results of Model (4). In all models, we first estimate the demand for full sample, and included multigroup analysis that allows us to compare the estimated coefficients between Groups 1, 2 and 3. This setting also lets us to examine whether the differences in the estimated coefficients are statistically significant. In equation form, we simultaneously estimate the Model (4) as following:

$$\boldsymbol{w} = \boldsymbol{A}\boldsymbol{\eta} + \boldsymbol{B}\boldsymbol{x} + \boldsymbol{\epsilon},\tag{1}$$

$$\boldsymbol{y} = \boldsymbol{\Lambda}\boldsymbol{\eta} + \boldsymbol{\nu}, \tag{2}$$

where  $w(2 \times 1)$  is a vector of dependent variables of WTB/FAV value,  $\eta(6 \times 1)$  is a vector of 6 latent constructs,  $x(14 \times 1)$  is a set of 14 explanatory variables,  $y(27 \times 1)$  is a vector of 27 indicator variables for estimating latent variables,  $A(2 \times 6)$ ,  $B(2 \times 14)$ , and  $\Lambda(27 \times 6)$ are matrices of unknown parameters to be estimated, and  $\epsilon(2 \times 1)$  and  $\nu(27 \times 1)$  are error terms. Equation (1) is called a structural equation, which describes the relationship between the latent and dependent variables, while equation (2) is a measurement equation, which is used for estimating the factor loading matrix  $\Lambda$  and predicting the score of latent variables  $\eta$ . Specifically, equation (1) can also be written as follows:

$$\begin{bmatrix} WTB_{i} \\ Value_{i} \end{bmatrix} = \begin{bmatrix} \alpha_{11} & \alpha_{12} & \alpha_{13} & \alpha_{14} & \alpha_{15} & \alpha_{16} \\ \alpha_{21} & \alpha_{22} & \alpha_{23} & \alpha_{24} & \alpha_{25} & \alpha_{26} \end{bmatrix} \begin{bmatrix} SP_{i} \\ ND_{i} \\ FE_{i} \\ MR_{i} \\ EP_{i} \\ EN_{i} \end{bmatrix} + \begin{bmatrix} \beta_{10} & \beta_{11} & \beta_{1x}' \\ \beta_{20} & 0 & \beta_{2x}' \end{bmatrix} \begin{bmatrix} 1 \\ Value_{i} \\ x_{i} \end{bmatrix} + \begin{bmatrix} \epsilon_{1i} \\ \epsilon_{2i} \end{bmatrix}$$
(3)

where *i* refers to each individual and  $\alpha$  presents the correlation between each latent construct and WTB/FAV value, which is our main focus. Thus  $WTB_i$  and  $Value_i$  refers to the WTB and FAV values of individual *i*, respectively. The parameters  $\beta_{10}$  and  $\beta_{20}$  are intercepts,  $\beta_{11}$  is the correlation between WTB and FAV value, and  $\beta_{1x}$  and  $\beta_{2x}$  show the correlation between WTB/FAV value and a set of 12 individual characteristics  $x_i$ . To ensure better model fit, we assume that some of the error terms associated with indicator variables are correlated. Hypothesizing a correlation between these error terms can improve our model's ability to explain the data.

## **5** Results

In this section, we first explain our results in general in Subsection 5.1, then demonstrate the importance of social support in Subsection 5.2.

#### 5.1 Main Result

In sum, our models provide suggestive evidence of the importance of social support in the WTB and FAV value. We report the results of Models (1) to (2) in Table 4 and Models (3) to (4) in Table 5 and 6. As mentioned in Section 4, the difference between Models (1) and (2) and Models (3) to (4) is whether we include the correlation between FAV value and WTB. We divide our results tables into three panels: Panel A displays the estimated coefficients of the latent variables, Panel B presents the estimation results of the socioe-conomic variables, and Panel C shows the model fit. In Table 4, the first and fifth columns show the estimation results for full sample, the second and sixth column shows the results of Group 1, the third and seventh columns display the estimation results for Group 3. For all models, we use standardized coefficients to permit comparison between the magnitudes of coefficients in all types of models, which is frequently used in quantitative studies to reflect the relative importance of explanatory variables within a model ([21]).

We do not observe substantial differences in model coefficients across Models (1) to (4). Thus, we choose to focus on Model (4) with the full specification when interpreting the results. The results of the measurement equation are shown in Table A3 in the Appendix. In this section, we briefly explain the results and further discuss on the implications in Subsection 6.1.

**Notes on Interpretation.** Before proceeding to interpreting the results, we want to clarify that people can have different combinations of latent variables. For instance, people can have high levels of both 'FE (Fear)' and 'MR (Merit)'. For example, concerns about accidents are negatively correlated with FAV value, and this does not indicate that a person with high levels of 'FE' does not appreciate the benefits of FAVs. Potential consumers would express appreciation for the benefits of FAVs given the results on the coefficients of 'MR.' Our result shows the changes in WTB and FAV value following one-standard-deviation increases in a latent variable, holding the other latent variables fixed. We focus on the comparison across the groups rather than explaining the results by groups. Overall, we find that the estimated coefficients of full samples are not substantially different from Group 1, and this is quite natural given that experiencing a severe level of disasters which are likely to be correlated to the loss of driving capability, is not common.

**Social Support (SP)** The results on social support are worthy of discussion. First, social support shows positive and statistically significant coefficients across all types of models and dependent variables, suggesting that social support is highly likely to encourage people to adopt/appreciate FAVs. Second, most of our results suggest that disaster victims show a substantially higher level of the social support coefficient than those who belong to Group 1 and full sample, indicating that disaster victims are more likely to regard social support as vital than those who do not experience disasters. Our results indeed provide suggestive evidence on the need to urge social support. We further discuss the policy-relevant implications of social support in Section 6.2. This observation is consistent with anecdotes from documents published by the Japanese government, indicating the support from local authorities and family is vital for achieving inclusive transportation.<sup>12</sup>

**Natural Disaster (ND)** The coefficients of ND are all positive and statistically significant, indicating that fear of natural disasters is likely to discourage people from choosing/appreciating FAVs. Note that ND is higher if people are *not* afraid of disasters. Thus, a positive coefficient implies that people would adapt/appreciate FAVs if they are not frightened of natural disasters. Note that the estimated coefficient of ND, in terms of FAV value, is slightly smaller in Group 2(0.038) and Group 3 (0.044) than in Group 1 (0.055)

<sup>&</sup>lt;sup>12</sup>We refer to the Japanese Cabinet's Guideline entitled "Social Principles of Human-Centric AI" (2020).

and full sample (0.055). On the other hand, for WTB, the estimated coefficient of ND would be more significant for those who belong to Group 2 (0.074) and Group 3 (0.082) than to Group 1 (0.055) and entire samle (0.060). This result suggests that the effect of ND is more likely to prevent disaster victims from choosing FAVs (as expressed in WTB), while it does not prevent disaster victims from appreciating FAVs (as depicted by FAV value). Nonetheless, the differences between coefficients across groups were not substantial.

**Fear (FE) and Merit (MR)** As expected, merit would encourage and accidents would discourage people from choosing/appreciating FAVs. These results are reasonable in the sense that people who appreciate the benefits of using FAVs would have higher WTB and FAV value, and those who fear potential accidents would not be more likely to purchase FAVs or appreciate FAVs than those who do not fear FAVs. Similar to ND, we do not find substantial differences between groups and WTB/FAV value.

**Environmental Concerns ((EP) and (EN))** EP (which stands for Pollution) shows insignificant coefficients in WTP, it displays a positive coefficient of 0.022 in Group 1's WTB, a negative coefficient of -0.047 in Group 2, and an insignificant coefficient in Group 3. On the other hand, EN (which stands for Nature) shows positive coefficients across groups, both in WTB and FAV value.

**Socioeconomic Variables** Regarding the remaining parameters, as expected and consistent with the previous works on FAV demand, socioeconomic attributes mostly exhibit statistically significant coefficients, implying that these characteristics also play a role in FAV demand. We also find some interesting results for the socioeconomic variables: 1) women and older respondents present high FAV value but lower WTB. 2) Car ownership presents a negative coefficient for FAV value and a positive coefficient for WTB. This result is explained by previous works ([1], [89], [79]), which report that drivers enjoy or are satisfied with the sense of driving themselves and thus may not be attracted

by FAVs. This result indicates that there should be more policy-relevant efforts to encourage drivers to appreciate the benefits of FAVs. 3) (Plug-in) hybrid cars show positive coefficients, indicating that those who purchase such cars are likely to be attracted by FAVs, which is consistent with previous works ([6], [70]). Other socioeconomic variables, particularly those correlated with commuting behavior, were either insignificant or inconsistent across groups, indicating weak correlation between them and the WTB/FAV value for FAVs.

**Model Fit** According to the goodness-of-fit indices shown at the bottom of the table, in general, the models fit the data modestly well. The acceptable range of RMSEA is < 0.08, those of CFI and SRMR are < 0.90 and <0.05, respectively. ([65], [37]). In our model, the values of RMSEA, CFI and SRMR are generally within or near each variable's acceptable range.

**WTB and FAV Value** Throughout Models (3) and (4), we find positive correlations between WTB and FAV value of approximately 0.3, indicating the 'overall' trend that people with a high level of WTB are likely to have higher perceived FAV Value and vice versa.

The results above indicate the importance of social support as well as other latent variables. However, it is difficult to determine whether the differences across groups are statistically significant, which inspires us to test the group heterogeneity of the estimated coefficients in Subsection 5.2 below.

	Model (1) (WTB)				Model (2) (FAV value)			
	All	Group 1	Group 2	Group 3	All	Group 1	Group 2	Group 3
Panel A: Latent Varia	bles							
SP	0.069***	0.095***	0.088***	0.152***	0.101***	0.060***	0.079***	0.126***
	(.004)	(.005)	(.014)	(.016)	(.004)	(.005)	(.014)	(.016)
ND	0.076***	0.055***	0.038***	0.044**	0.055***	0.071***	0.086***	0.095***
	(.004)	(.004)	(.014)	(.016)	(.004)	(.004)	(.013)	(.016)
FE	-0.082***	-0.077***	-0.103***	-0.113***	-0.083***	-0.077***	-0.102***	-0.106***
	(.008)	(.009)	(.024)	(.032)	(.008)	(.009)	(.024)	(.033)
MR	0.215***	0.142***	0.150***	0.146***	0.143***	0.217***	0.203***	0.231***
ED	(.008)	(.009)	(.025)	(.033)	(.008)	(.009)	(.025)	(.033)
EP	0.013	0.011	-0.027	-0.0388	0.006	0.026	-0.048	-0.052
FN	0.009)	0.009)	0.026)	0.106**	0.061***	(.009) 0 0/0***	(.026) 0 112***	0.101**
	(009)	(009)	(026)	(034)	(009)	(009)	(026)	(034)
Panel B: Socioecono	mic Variable	es	(.020)	(.034)	(.003)	(.003)	(.020)	(.034)
Female	0.041***	0.064***	0.053***	0.033**	-0.059***	-0.039***	-0.034***	-0.048**
	(.004)	(.005)	(.013)	(.017)	(.004)	(.005)	(.013)	(.017)
ln(Income)	0.095***	0.109***	0.103***	0.110***	0.108***	0.100***	0.079***	0.065***
1 (1 )	(.004)	(.005)	(.013)	(.017)	(.004)	(.005)	(.013)	(.017)
ln(Age)	-0.116***	0.090***	0.093***	0.033*	0.086***	-0.006	-0.020	-0.059***
Howeehold Size	(.004)	(.005)	(.013)	(.017)	(.004)	(.005)	(.013)	(.017)
Household Size	-0.011	-0.020	-0.0484	-0.027	-0.024	-0.008	-0.018	-0.014
Preschooler	0.016***	(.005) 0 029***	0.012)	0.0198	0.004)	0.005)	0.012)	-0.003
riesenooiei	(004)	(005)	(012)	(016)	(004)	(005)	(026)	(016)
Car Ownership	-0.044***	-0.042***	-0.037***	-0.029*	-0.040***	0.045***	0.031***	0.027*
ouronnerenp	(.004)	(.004)	(.012)	(.016)	(.004)	(.004)	(.012)	(.016)
Hybrid	0.061***	0.043***	0.042***	0.030**	0.043***	0.060***	0.059***	0.068***
·	(.004)	(.004)	(.011)	(.015)	(.004)	(.004)	(.011)	(.015)
Plug-in Hybrid	0.014**	0.009**	0.002	-0.008	0.008**	0.014***	0.018	-0.009
	(.004)	(.004)	(.011)	(.015)	(.004)	(.004)	(.011)	(.015)
ln(Commute Time)	-0.024***	-0.030***	-0.028	-0.039*	-0.031***	0.025***	0.01	0.042
	(.007)	(.008)	(.022)	(.028)	(.007)	(.008)	(.012)	(.027)
Long Commute	-0.005	0.003	-0.009	0.02	0.004	-0.008*	0.004	0.008
	(.004)	(.004)	(.012)	(.016)	(.004)	(.004)	(.012)	(.016)
Short Commute	-0.025	-0.025***	-0.035**	-0.014	-0.024	-0.003	-0.013	0.0328
ln(Commute Cost)	(.006) 0 027***	(.006)	(.017)	(.023) 0 032*	(.006)	(.006) 0 025***	(.017) 0 0182	(.022)
	0.027	-0.003	0.000	0.032	-0.001	0.025	0.0102	0.031
Observations	69.391	57.105	7.853	4.433	69.391	57.105	7.853	4.433
Panel C: Model Fit	50,001	51,100	.,000	1,100	00,001	51,100	1,000	1,100
RMSEA	0.045		0.044		0.045		0.044	
CFI	0.916		0.913		0.916		0.913	
SRMR	0.05		0.06		0.05		0.06	

Table 4: Estimation Results of Models (1) and (2)

Note: Standard errors in parentheses. \* p<0.1, \*\* p<0.05, \*\*\* p<0.01.

	Model (3) (WTB)								
	All	Group 1	Group 2	Group 3	All	Group 1	Group 2	Group 3	
FAV value		-	-	-	0.337***	0.337***	0.327***	0.338***	
					(.003)	(.004)	(.010)	(.013)	
Panel A: Latent Varia	bles								
SP	0.101***	0.095***	0.088***	0.153***					
ND	(.004) <b>0.055***</b>	(.005) <b>0.055***</b>	(.013) <b>0.038***</b>	(.023) <b>0.044***</b>					
FE	(.004) -0.083***	(.004) - <b>0.077***</b>	(.014) -0.103***	(.016) - <b>0.113***</b>					
MR	(.008) <b>0.143***</b>	(.009) <b>0.142***</b>	(.024) <b>0.150***</b>	(.032) <b>0.147***</b> *					
EP	(.008) <b>0.006</b>	(.009) <b>0.011</b>	(.025) - <b>0.027</b>	(.033) -0.039					
EN	(.009) <b>0.061***</b>	(.009) <b>0.05</b> 4***	(.026) <b>0.073***</b>	(.035) <b>0.106**</b>					
	(.009)	(.009)	(.026)	(.035)					
Panel B: Socioecono	mic Variable	es	0.052***	0 000** *					
Female	0.059***	0.064***	0.053***	0.033*** *					
ln(Income)	0.108***	0.109***	0.103 <sup>***</sup>	0.110***					
ln(Age)	(.004) <b>0.086***</b>	(.005) <b>0.090***</b>	(.013) <b>0.093***</b>	(.017) <b>0.0328</b> *					
Household Size	(.004) -0.240***	(.005) - <b>0.020***</b>	(.013) - <b>0.048***</b>	(.017) - <b>0.027</b>					
Preschooler	(.004) <b>0.030***</b>	(.005) <b>0.029***</b>	(.012) <b>0.044***</b>	(.016) <b>0.0197</b>					
	(.004)	(.005)	(.012)	(.016)					
Car Ownership	-0.040***	-0.042***	-0.037***	-0.029*					
Hybrid	0.043***	0.043***	0.042***	0.030*					
Dhara in Hadanid	(.004)	(.004)	(.011)	(.015)					
Plug-III Hybrid	0.008	0.009	0.002	-0.008					
ln( <b>Commute Time</b> )	-0.031***	-0.030***	-0.029	-0.0394					
	(.007)	(.008)	(.022)	(.028)					
Long Commute	0.004	0.003	-0.009	0.0205					
0	(.004)	(.004)	(.012)	(.016)					
Short Commute	-0.026***	-0.025***	-0.035*	-0.014					
	(.006)	(.006)	(.017)	(.022)					
ln(Commute Cost)	-0.001	-0.005	0.006	0.0316					
	(.006)	(.006)	(.015)	(.019)					
Observations	69,105	57,105	7,853	4,433					
Panel C: Model Fit	0.045			_					
RMSEA	0.045			0.04	4				
CFI	0.913			0.91	.3				
SKMR	0.06	* ^	1 ** 0.02	0.0	b				
Note: Standard error	Note: Standard errors in parentheses. * $p < 0.1$ , ** $p < 0.05$ , *** $p < 0.01$ .								

	A 11	~ ·		_				Model (4) (WTB))				
	AII	Group I	Group 2	Group 3	All	Group 1	Group 2	Group 3				
FAV value					0.304***	0.305***	0.299***	0.290***				
					(.004)	(.004)	(.011)	(.014)				
Panel A: Latent Varia	bles											
SP	0.101***	0.095***	0.088***	0.153***	0.039***	0.031***	0.053***	0.083***				
	(.004)	(.005)	(.014)	(.023)	(.004)	(.005)	(.013)	(.016)				
ND	0.055***	0.055***	0.038***	$0.044^{***}$	0.060***	0.055***	$0.074^{***}$	0.082***				
	(.004)	(.004)	(.014)	(.016)	(.004)	(.004)	(.013)	(.015)				
FE	-0.084***	-0.078***	-0.104***	-0.113***	-0.057***	-0.053***	-0.071***	-0.073**				
	(.008)	(.009)	(.024)	(.032)	(.007)	(.008)	(.023)	(.031)				
MR	$0.144^{***}$	0.143***	0.151***	$0.147^{***}$	0.172***	0.173***	0.158***	0.189***				
	(.009)	(.009)	(.025)	(.033)	(.008)	(.009)	(.024)	(.032)				
EP	0.011	0.011	-0.003	-0.039	0.011	0.022**	-0.047**	-0.041				
	(.009)	(.009)	(.026)	(.035)	(.008)	(.009)	(.025)	(.033)				
EN	0.061***	0.055***	0.074***	0.106**	0.055***	0.033***	0.090***	0.076**				
	(.009)	(.009)	(.026)	(.035)	(.008)	(.009)	(.024)	(.033)				
Panel B: Socioeconor	mic Variables	6 0 0 1 * * *	0.050***	0.000**	0.050***	0.050***	0.040***	0.050***				
Female	0.059***	0.064***	0.053***	0.033**	-0.059***	-0.059***	-0.049***	-0.058***				
1 (1	(.004)	(.005)	(.013)	(.017)	(.004)	(.005)	(.012)	(.016)				
In(Income)	0.108	0.109****	0.103	0.110	0.062	0.067****	0.049	0.033**				
1 (1	(.004)	(.005)	(.013)	(.017)	(.004)	(.005)	(.012)	(.016)				
In(Age)	0.086***	0.090***	0.093***	0.0328*	-0.038***	-0.034***	-0.048***	-0.068***				
Household Size	(.004)	(.005)	(.013)	(.017)	(.004)	(.005)	(.011)	(.016)				
Household Size	-0.024	-0.020****	-0.049	-0.027	-0.009****	-0.002	-0.004	-0.006				
Dreachaolar	(.004) 0 020***	(.005) 0 0 0 0 0 * * *	(.012) 0 044***	(.016)	(.004) 0 007**	(.004)	(.012)	(.016)				
Preschoolei	0.050	0.029	0.044	0.0197	0.007	0.009	0.002	-0.000				
Car Ownorship	(.004) 0 0/02***	(.005) 0 042***	(.015) 0 027***	(.016) 0 020*	(.004) 0 056***	(.004) 0 058***	(.012) 0 042***	(.015) 0 0354**				
CarOwnership	-0.0402	-0.043	-0.037	-0.025	0.030	0.030	0.042	0.0334				
Hybrid	(.004) 0 0/2***	(.004) 0 0/13***	(.012) 0 0/2***	0.016)	(.004) 0 0/08***	(.004) 0 047***	0.011)	(.015) 0 059***				
Tiybild	0.045	0.045	0.042	0.030	0.040	0.047	0.040	0.000				
Plug_in Hybrid	0.004)	0.004)	-0.002	-0.008	0.011**	0.011***	0.017	-0.006				
i iug-iii i iybiiu	(004)	0.005	-0.002	-0.000	0.011	0.011	(011)	-0.000				
ln(Commute Time)	-0 031***	-0 030***	-0.028	-0 0394	0 033***	0 034***	0.018*	0.054**				
m(commute rime)	(007)	( 008)	(022)	( 028)	(003)	(007)	(015)	(026)				
Long Commute	0.004	0.003	-0.001	0.0205	-0.006	-0.008**	0.004	0.002				
Long Commute	(004)	(004)	(012)	(016)	(004)	(004)	(011)	(016)				
Short Commute	-0.026***	-0.025***	-0.035*	-0.014	0.005	0.005	-0.003	0.037*				
Short Commute	(006)	(006)	(016)	( 022)	( 005)	(006)	(016)	(021)				
ln(Commute Cost)	-0.002	-0.005	0.006	0.0316	0.027***	0.026***	0.018	0.041***				
	(.005)	(.006)	(.015)	(.019)	(.005)	(.005)	(.015)	(.018)				
Observations	69.105	57.105	7.853	4,433	69.105	57.105	7.853	4,433				
Panel C: Model Fit			.,===	-,	,	,=	.,	-,				
RMSFA	0.045				0.044							
CFI	0.916				0.044							
SRMR	0.05				0.01							
Note: Standard error	s in parenthe	eses. * p<0.1	, ** p<0.05.	*** p<0.01.	0.00							
NDFEMREPENPanel B: SocioeconorFemaleIn(Income)In(Age)Household SizePreschoolerCar OwnershipHybridPlug-in HybridIn(Commute Time)Long CommuteShort CommuteIn(Commute Cost)ObservationsPanel C: Model FitRMSEACFISRMRNote: Standard error	(.004) 0.055*** (.004) -0.084*** (.008) 0.144*** (.009) 0.011 (.009) 0.061*** (.009) 0.061*** (.009) 0.059*** (.004) 0.086*** (.004) 0.024*** (.004) 0.030*** (.004) 0.042*** (.004) 0.043*** (.004) 0.043*** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.004) 0.008** (.005) 69,105 0.05 5 in parenthe	(.005) $0.055^{***}$ (.004) $-0.078^{***}$ (.009) $0.143^{***}$ (.009) 0.011 (.009) $0.055^{***}$ (.009) $0.064^{***}$ (.005) $0.090^{***}$ (.005) $0.020^{***}$ (.005) $0.020^{***}$ (.005) $0.020^{***}$ (.005) $0.020^{***}$ (.005) $0.020^{***}$ (.005) $0.020^{***}$ (.004) $0.043^{***}$ (.004) $0.009^{***}$ (.004) $0.009^{***}$ (.004) $0.009^{***}$ (.004) $0.003^{***}$ (.008) $0.003^{***}$ (.006) $57,105^{***}$ Esees. * $p < 0.1$	(.014) 0.038*** (.014) -0.104*** (.024) 0.151*** (.025) -0.003 (.026) 0.074*** (.013) 0.103*** (.013) 0.093*** (.013) 0.093*** (.013) -0.049*** (.012) 0.044*** (.015) -0.037*** (.011) -0.002 (.012) -0.028 (.022) -0.001 (.012) -0.035* (.016) 0.006 (.015) 7,853	(.023) 0.044*** (.016) -0.113*** (.032) 0.147*** (.033) -0.039 (.035) 0.106** (.035) 0.106** (.017) 0.0328* (.017) 0.0328* (.017) 0.0328* (.017) 0.027 (.016) -0.029* (.016) -0.029* (.015) -0.008 (.015) -0.008 (.015) -0.0394 (.028) 0.0205 (.016) -0.0394 (.028) 0.0205 (.016) -0.014 (.022) 0.0316 (.019) 4,433	(.004) 0.060*** (.004) -0.057*** (.007) 0.172*** (.008) 0.011 (.008) 0.055*** (.004) -0.038*** (.004) -0.009*** (.004) 0.007** (.004) 0.007** (.004) 0.0056*** (.004) 0.048*** (.004) 0.033*** (.004) 0.033*** (.004) 0.033*** (.004) 0.033*** (.004) 0.033*** (.004) 0.033** (.004) 0.033** (.005) 0.027*** (.005) 0.027***	(005) $0.055^{***}$ (004) $-0.053^{***}$ (008) $0.173^{***}$ (009) $0.022^{**}$ (009) $0.033^{***}$ (009) $-0.059^{***}$ (005) $-0.034^{***}$ (005) -0.002 (004) $0.009^{***}$ (004) $0.047^{***}$ (004) $0.034^{***}$ (004) $0.034^{***}$ (004) $0.034^{***}$ (004) $0.034^{***}$ (004) $0.034^{***}$ (004) $0.034^{***}$ (004) $0.034^{***}$ (004) 0.005 (004) 0.005 (006) $0.026^{***}$ (005) 57,105	(.013) 0.074*** (.013) -0.071*** (.023) 0.158*** (.024) -0.047** (.025) 0.090*** (.024) -0.049*** (.012) 0.049*** (.011) -0.048*** (.011) -0.004 (.012) 0.042*** (.011) 0.046*** (.011) 0.046*** (.011) 0.046*** (.011) 0.046*** (.011) 0.017 (.011) 0.018* (.015) 0.003 (.016) 0.018 (.015) 7,853	(.016) 0.082*** (.015) -0.073** (.031) 0.189*** (.032) -0.041 (.033) 0.076** (.033) -0.058*** (.016) -0.068*** (.016) -0.006 (.016) -0.006 (.015) 0.0354** (.015) 0.059*** (.019) -0.006 (.015) 0.059*** (.019) -0.006 (.015) 0.059*** (.019) -0.006 (.015) 0.059*** (.019) -0.006 (.015) 0.059*** (.019) -0.006 (.015) 0.059*** (.019) -0.006 (.015) 0.059*** (.019) -0.006 (.015) 0.059*** (.019) -0.006 (.015) 0.059*** (.019) -0.006 (.015) 0.059*** (.019) -0.006 (.015) 0.054** (.021) 0.0037* (.021) 0.041*** (.018) 4,433				

#### Table 6: Estimation Results of Model (4) Model (4) (FAV value)

#### 5.2 Supplementary Results

To determine whether the difference in estimated coefficients across groups is statistically significant, focusing on the latent variables, we test for invariance of the coefficients of each latent variable across groups. We conduct Wald tests to indicate whether we can reject the null hypothesis that the coefficients estimated in Model (4) do not differ across groups. Rejecting the null hypothesis implies the existence of group heterogeneity. Table 7 shows the results of the Wald tests and is divided into two panels: Panel A shows the Wald test results for FAV value, and Panel B presents the Wald test results for WTB. The first column provides the chi-squared statistics on the differences between Groups 1 and 2. Analogously, the second and third columns show the results for Groups 2 and 3 and Groups 1 and 3, respectively.

The results reveal that social support is the only latent with coefficients that differ substantially and significantly across groups. This trend is more substantial for FAV value because there are no statistically significant differences in coefficients across groups other than social support. We find additional evidence that social support has a significantly stronger correlation with WTB for Group 3 than Group 1 and 2. This is because comparing Group 1 to 3 shows higher and statistically significant coefficients (13.524 for FAV value, 9.803 for WTB) while comparing Group 1 to 2 does not yield statistically significant coefficients. This result allows us to interpret that people belonging to Group 3 recognize social support as a critical factor. Other than social support, the coefficients of environmental concerns (pollution and nature) differ significantly between Groups 1 and 2. The difference in the coefficients of ND and EP between Groups 1 and 3 are also significant, but the significance is relatively weak.

Finally, concluding on the importance of social support requires us to check the correlations between social support and other latent constructs. Suppose that social support is not highly correlated with other latents. In that case, we could conclude that social support is a unique and independent factor that is vital for inspiring disaster victims to choose FAVs. Table 8 shows the correlation matrix among the predicted scores of the la-

29

tent variables. We do not find any latents that are closely correlated with social support.

Other than social support, the correlation between pollution (EP) and nature (EN) is the highest among the six latent constructs (0.889), followed by Fear (FE) and Merit (MR), which shows a correlation of 0.821. This result implies that those with a high level of EP (FE) have higher EN (MR), which is reasonable.

Table 7: Difference in Coefficients across Groups								
	Group 1 vs. 2	Group 2 vs. 3	Group 1 vs. 3					
Panel A: FAV value								
SP	0.519	12.756***	13.524***					
ND	1.629	0.286	0.228					
FE	0.839	0.293	1.383					
MR	0.013	0.096	0.102					
EP	0.261	0.794	1.819					
EN	0.383	0.946	2.393					
Panel B: WTB								
SP	2.515	2.035	9.803***					
ND	2.17	0.15	3.025*					
FE	0.552	0.012	0.375					
MR	0.329	0.756	0.189					
EP	7.099***	0.015	3.382*					
EN	4.886**	0.114	1.585					

*Note*: Each value shows the chi-squared statistic with significance obtained from the Wald test.\* p<0.1, \*\* p<0.05, \*\*\* p<0.01

Table 8: Corre	elation	between	Predi	cted Score	es of l	Latent Variables
	SP	ND	FE	MR	EP	EN

	51	ND	ГĽ	IVIII	LI	LIN
SP	1					
ND	0.049	1				
FE	-0.018	0.150	1			
MR	0.016	0.076	0.821	1		
EP	0.107	0.284	0.284	0.286	1	
EN	0.133	0.095	0.204	0.189	0.889	1

# 6 Discussion and Policy Implications

Our results presented in the previous sections demonstrate the importance of social support in making FAVs inclusive. In this section, we first discuss our results in general in Subsection 6.1, then illuminate the importance of social supports in Subsection 6.2.

#### 6.1 Result Implications

We answer whether FAVs can be an inclusive transport mode by examining how FAV adoption and appreciation are shaped, focusing on disaster victims. To do so, we consider individual-level demographic characteristics and individual-level psychometric attributes of social supports, fear of a natural disaster, natural environment conservation, pollution reductions, accident concerns, and merits. By including such characteristics, this study enables policymakers to move beyond simply observing the growth of FAV adoption patterns to actively directing the adoption path and making it inclusive. Our results inspire the design of effective policy instruments and information campaigns that appeal to disaster victims. Therefore, our findings provide clear, practical contributions.

This study discovers that social support is crucial for achieving inclusive transportation in the context of FAVs: Social support critically encourages disaster victims to choose and appreciate FAVs. This finding is consistent with previous works showing that family encouragement is a suitable stimulus for socially vulnerable people to accept new technology ([35]). The significance of the group heterogeneity coefficient provides further support for our findings. Our result suggests that social support is the only latent factor that exhibits statistically significant group heterogeneity coefficients in both WTB and FAV value.

Second, while previous works reveal that a higher level of environmental concern motivates FAV adoption, interestingly, we find group heterogeneity in environmental concerns where the implications differ according to the type of environmental concern (natural environmental preservation and pollution reduction). Our result emphasizes the importance of separately identifying environmental awareness according to context. One possible explanation for this difference relies on the characteristics of disaster victims. Recall from the literature that disaster victims are interested in whether choosing FAVs would reduce or prevent damage from natural disasters. Activities related to pollution reduction are not directly connected to disaster damage mitigation (and are therefore less attractive to disaster victims). However, pollution reduction activities might be associated with direct environmental damage alleviation (and thus be attractive for the people without disaster experiences). Future studies can explore the detailed reasons for the differences and group heterogeneity.

Third, our result shows that the concerns about possible accidents and merits are significant predictors of FAV adoption. However, we do not find group heterogeneity in their estimated coefficients. Nonetheless, such a finding does not indicate that the accidents and merits are not interconnected to disaster victims' FAV adoption but shows that disaster experience does not shift the underlying importance of both factors.

#### 6.2 Social Support and Policy Implications

This study extends the work of [30], [48] that incorporates the effect of social interaction into the choice model. Specifically, it contributes by 1) examining disaster victims, 2) focusing on WTB and FAV value, 3) including other latent variables associated with disasters, accidents, merits, and environmental concerns, and 4) scrutinizing the group heterogeneity of social support with respect to disaster experience. Through such additional contributions, we confirm the significance of social support.

Our findings provide suggestive policy implications for policymakers worldwide who are keen to make FAVs inclusive. On the one hand, the interaction between WTB and social support suggests that encouraging social support through the government and local authorities establishing and publishing guidelines on FAVs would motivate widespread adoption of FAVs by disaster victims. On the other hand, an increase in social support can motivate disaster victims to perceive FAVs as beneficial for them. Thus, extending the findings of previous works highlighting the importance of social support in driving/vehicle choices ([82], [26]), this study provides additional implications by reconciling the findings from social support to fear of natural disasters, which is likely to hinder FAV adoptions/appreciations, with a statistically significant group heterogeneity coefficient for WTB.

Providing a governmental (or local institutional) guidelines on the prior knowledge of FAVs would significantly alleviate the negative influence of concerns associated with the locus of losing control. Most of the obstacles that hinder disaster victims from choosing FAVs derive from the fear of 'losing control,' particularly in times of accidents, which thus renders them more fearful than others of new technology ([7], [63]). In that sense, our study reaffirms the findings of the past studies that have noted the impact of prior knowledge on attitudes toward new kinds of vehicles (e.g., electric cars) and on intentions to use them ([73]). Such a finding would apply to both disaster victims and people without disaster experiences.

Family and peer support can encourage disaster victims to adapt to new technologies or return to driving, according to previous medical studies on traumatic symptoms ([62]). Rather than hastily pushing disaster victims to adopt and appreciate FAVs, gradually letting them adopt the usage of FAVs through repeated short-term travel or simulationbased training with the support of family and friends would be desirable. Such a conclusion also implies that social support can work as a stimulus for disaster victims to overcome the fear of disasters, which hinders them from choosing FAVs.

## 7 Concluding Remarks

Achieving inclusive transportation requires accommodating socially vulnerable people who are not able to drive unassisted. Focusing on FAVs and disaster victims, our results indicate that social support is pivotal in motivating disaster victims to adopt and appreciate FAVs. We find additional evidence on the relationship between other psychometric factors and socioeconomic factors with FAV demand.

While our study offers several advantages over previous studies, it also has several

important limitations and presents avenues for future work. First, our research adopts SEM, which contains a possible concern of reverse causality. For example, a disaster victim who had always wished to purchase an FAV might overestimate the merits of FAVs and underestimate the risks of accidents. Such reverse causality might be addressed by using an instrumental variable (IV) approach. For example, IVs that control for the traits of early adopters or those with fixed demands can be employed. Unfortunately, we lack the variables necessary to differentiate them.

Although we use data from Japan because doing so allows us to secure a sufficient number of disaster victims for the sample, we believe that our empirical framework can be extended to a global context, as many countries are keen on achieving inclusive transportation and encourage people to choose FAVs. Ultimately, by illuminating the importance of social support for FAV choices, our study suggests the importance of providing prior knowledge through institutional guidelines and enticing disaster victims to choose FAVs by increasing support from family and friends. Doing so could allow FAVs to be inclusive in the future.

#### Contributions

Sunbin YOO: Conceptualization, Methodology, Formal analysis, Writing (Original Draft); Junya KUMAGAI: Conceptualization, Methodology, Writing (Review and Editing); Yuta KAWABATA: Writing (Review and Editing); Shunsuke MANAGI: Resources and Supervision.

# References

- Ransford A. Acheampong and Federico Cugurullo. Capturing the behavioural determinants behind the adoption of autonomous vehicles: Conceptual frameworks and measurement models to predict public transport, sharing and ownership trends of self-driving cars. <u>Transportation Research Part F:</u> Traffic Psychology and Behaviour, 62:349–375, 2019.
- [2] Mohammad Aghababaei, Seosamh B. Costello, and Prakash Ranjitkar. Transportation impact assessment following a potential alpine fault earthquake in new zealand. <u>Transportation Research Part D:</u> Transport and Environment, 87:102511, 2020.
- [3] M. Aklin, P. Bayer, S.P. Harish, and J. Urpelainen. Economics of household technology adoption in developing countries: Evidence from solar technology adoption in rural india. <u>Energy Economics</u>, 72:35–46, 2018.
- [4] Adriano Alessandrini, Andrea Campagna, Paolo Delle Site, Francesco Filippi, and Luca Persia. Automated vehicles and the rethinking of mobility and cities. <u>Transportation Research Procedia</u>, 5:145– 160, 2015. SIDT Scientific Seminar 2013.
- [5] Jonn Axsen and Kenneth S. Kurani. Interpersonal influence in the early plug-in hybrid market: Observing social interactions with an exploratory multi-method approach. <u>Transportation Research</u> Part D: Transport and Environment, 16(2):150–159, 2011.
- [6] Prateek Bansal and Kara M. Kockelman. Forecasting americans' long-term adoption of connected and autonomous vehicle technologies. <u>Transportation Research Part A: Policy and Practice</u>, 95:49– 63, 2017.
- [7] Roger Bennett, Rohini Vijaygopal, and Rita Kottasz. Willingness of people with mental health disabilities to travel in driverless vehicles. Journal of Transport & Health, 12:1–12, 2019.
- [8] Alexandra M. Boggs, Ramin Arvin, and Asad J. Khattak. Exploring the who, what, when, where, and why of automated vehicle disengagements. Accident Analysis & Prevention, 136:105406, mar 2020.
- [9] Nicolás C. Bronfman, Pamela C. Cisternas, Esperanza López-Vázquez, and Luis A. Cifuentes. Trust and risk perception of natural hazards: implications for risk preparedness in chile. <u>Natural Hazards</u>, 81(1):307–327, Mar 2016.
- [10] Pei Nen Esther Chee, Yusak O. Susilo, and Yiik Diew Wong. Determinants of intention-to-use first-/last-mile automated bus service. <u>Transportation Research Part A: Policy and Practice</u>, 139:350–375, 2020.
- [11] Chienfei Chen, Hannah Nelson, Xiaojing Xu, Gregory Bonilla, and Nicholas Jones. Beyond technology adoption: Examining home energy management systems, energy burdens and climate change perceptions during covid-19 pandemic. <u>Renewable and Sustainable Energy Reviews</u>, 145:111066, 2021.

- [12] Juan de Oña and Rocio de Oña. Quality of service in public transport based on customer satisfaction surveys: A review and assessment of methodological approaches. <u>Transportation Science</u>, 49(3):605– 622, 2015.
- [13] Shuchisnigdha Deb, Lesley Strawderman, Daniel W. Carruth, Janice DuBien, Brian Smith, and Teena M. Garrison. Development and validation of a questionnaire to assess pedestrian receptivity toward fully autonomous vehicles. <u>Transportation Research Part C: Emerging Technologies</u>, 84:178– 195, nov 2017.
- [14] Xavier Demeulenaere. How challenges of human reliability will hinder the deployment of semiautonomous vehicles. Technological Forecasting and Social Change, 157:120093, 2020.
- [15] Xuan Di, Xu Chen, and Eric Talley. Liability design for autonomous vehicles and human-driven vehicles: A hierarchical game-theoretic approach. <u>Transportation Research Part C: Emerging</u> Technologies, 118:102710, sep 2020.
- [16] Shelly Etzioni, Ricardo A. Daziano, Eran Ben-Elia, and Yoram Shiftan. Preferences for shared automated vehicles: A hybrid latent class modeling approach. <u>Transportation Research Part C: Emerging</u> Technologies, 125:103013, 2021.
- [17] Koen Faber and Dea van Lierop. How will older adults use automated vehicles? assessing the role of avs in overcoming perceived mobility barriers. <u>Transportation Research Part A: Policy and Practice</u>, 133:353–363, 2020.
- [18] Daniel J. Fagnant and Kara Kockelman. Preparing a nation for autonomous vehicles: opportunities, barriers and policy recommendations. <u>Transportation Research Part A: Policy and Practice</u>, 77:167– 181, 2015.
- [19] Francesca Favarò, Sky Eurich, and Nazanin Nader. Autonomous vehicles' disengagements: Trends, triggers, and regulatory limitations. Accident Analysis & Prevention, 110:136–148, jan 2018.
- [20] Christos Gkartzonikas and Konstantina Gkritza. What have we learned? a review of stated preference and choice studies on autonomous vehicles. <u>Transportation Research Part C: Emerging Technologies</u>, 98:323–337, jan 2019.
- [21] Arthur S. Goldberger. Structural equation methods in the social sciences. <u>Econometrica</u>, 40(6):979–1001, 1972.
- [22] Chana J. Haboucha, Robert Ishaq, and Yoram Shiftan. User preferences regarding autonomous vehicles. Transportation Research Part C: Emerging Technologies, 78:37–49, may 2017.
- [23] Mohamed Mouloud Haddak, Marie Lefèvre, and Nathalie Havet. Willingness-to-pay for road safety improvement. Transportation Research Part A: Policy and Practice, 87:1–10, 2016.
- [24] Ziqiang Han, Xiaoli Lu, Elisa I. Hörhager, and Jubo Yan. The effects of trust in government on earthquake survivors' risk perception and preparedness in china. <u>Natural Hazards</u>, 86:437–452, dec 2016.

- [25] Emily W. Harville, Marni Jacobs, and Renée Boynton-Jarrett. When is exposure to a natural disaster traumatic? comparison of a trauma questionnaire and disaster exposure inventory. <u>PLOS ONE</u>, 10(4):1–11, 04 2015.
- [26] Eva Heinen. Identity and travel behaviour: A cross-sectional study on commute mode choice and intention to change. <u>Transportation Research Part F: Traffic Psychology and Behaviour</u>, 43:238–253, 2016.
- [27] Ming-Chou Ho, Daigee Shaw, Shuyeu Lin, and Yao-Chu Chiu. How do disaster characteristics influence risk perception? Risk Analysis, 28(3):635–643, 2008.
- [28] Johan Jansson, Thomas Pettersson, Andrea Mannberg, Runar Brännlund, and Urban Lindgren. Adoption of alternative fuel vehicles: Influence from neighbors, family and coworkers. Transportation Research Part D: Transport and Environment, 54:61–73, 2017.
- [29] Peng Jing, Liuyang Du, Yuexia Chen, Yuji Shi, Fengping Zhan, and Junping Xie. Factors that influence parents' intentions of using autonomous vehicles to transport children to and from school. <u>Accident</u> Analysis & Prevention, 152:105991, mar 2021.
- [30] Maria Kamargianni, Moshe Ben-Akiva, and Amalia Polydoropoulou. Incorporating social interaction into hybrid choice models. Transportation, 41(6):1263–1285, Nov 2014.
- [31] Shuqing Kang, Aupal Mondal, Aarti C. Bhat, and Chandra R. Bhat. Pooled versus private ride-hailing: A joint revealed and stated preference analysis recognizing psycho-social factors. <u>Transportation</u> Research Part C: Emerging Technologies, 124:102906, 2021.
- [32] Sebastian Kapser and Mahmoud Abdelrahman. Acceptance of autonomous delivery vehicles for lastmile delivery in germany – extending UTAUT2 with risk perceptions. <u>Transportation Research Part</u> C: Emerging Technologies, 111:210–225, feb 2020.
- [33] Takaaki Kato. Sharing experience by residents of past disaster sites: Effects of time from disasters and types of experience. International Journal of Disaster Risk Reduction, 61:102390, 2021.
- [34] Sherrie-Anne Kaye, Ioni Lewis, Sonja Forward, and Patricia Delhomme. A priori acceptance of highly automated cars in australia, france, and sweden: A theoretically-informed investigation guided by the tpb and utaut. Accident Analysis & Prevention, 137:105441, 2020.
- [35] Susan Kenyon, Glenn Lyons, and Jackie Rafferty. Transport and social exclusion: investigating the possibility of promoting inclusion through virtual mobility. Journal of Transport Geography, 10(3):207–219, 2002.
- [36] Pantelis Kopelias, Elissavet Demiridi, Konstantinos Vogiatzis, Alexandros Skabardonis, and Vassiliki Zafiropoulou. Connected and autonomous vehicles-environmental impacts-a review. <u>Science of</u> The Total Environment, 712:135237, 2020.

- [37] Junya Kumagai, Mihoko Wakamatsu, and Shunsuke Managi. Do commuters adapt to in vehicle crowding on trains? Transportation, 2020.
- [38] Yi-Wen Kung and Sue-Huei Chen. Perception of earthquake risk in taiwan: Effects of gender and past earthquake experience. Risk Analysis, 32(9):1535–1546, 2012.
- [39] Calvin Lee Kwan. Influence of local environmental, social, economic and political variables on the spatial distribution of residential solar pv arrays across the united states. <u>Energy Policy</u>, 47:332–344, 2012.
- [40] Wen-Tai Lai and Ching-Fu Chen. Behavioral intentions of public transit passengers—the roles of service quality, perceived value, satisfaction and involvement. Transport Policy, 18(2):318–325, 2011.
- [41] Jeffrey J. LaMondia, Daniel J. Fagnant, Hongyang Qu, Jackson Barrett, and Kara Kockelman. Shifts in long-distance travel mode due to automated vehicles: Statewide mode-shift simulation experiment and travel survey analysis. Transportation Research Record, 2566(1):1–11, 2016.
- [42] Zoe Le Hong and Naomi Zimmerman. Air quality and greenhouse gas implications of autonomous vehicles in vancouver, canada. <u>Transportation Research Part D: Transport and Environment</u>, 90:102676, 2021.
- [43] Dasom Lee and David J. Hess. Regulations for on-road testing of connected and automated vehicles: Assessing the potential for global safety harmonization. <u>Transportation Research Part A: Policy and</u> Practice, 136:85–98, 2020.
- [44] Timo Liljamo, Heikki Liimatainen, and Markus Pöllänen. Attitudes and concerns on automated vehicles. Transportation Research Part F: Traffic Psychology and Behaviour, 59:24–44, 2018.
- [45] Diyi Liu, Huibin Du, Frank Southworth, and Shoufeng Ma. The influence of social-psychological factors on the intention to choose low-carbon travel modes in tianjin, china. <u>Transportation Research</u> Part A: Policy and Practice, 105:42–53, 2017.
- [46] C.J Lonigan, M.P Shannon, A.J Finch, T.K Daugherty, and C.M Taylor. Children's reactions to a natural disaster: symptom severity and degree of exposure. <u>Advances in Behaviour Research and Therapy</u>, 13(3):135–154, 1991. Impact of Natural Disasters on children and families.
- [47] Jiaqi Ma, Xiaopeng Li, Fang Zhou, and Wei Hao. Designing optimal autonomous vehicle sharing and reservation systems: A linear programming approach. <u>Transportation Research Part C: Emerging</u> Technologies, 84:124–141, nov 2017.
- [48] Francesco Manca, Aruna Sivakumar, Nicolò Daina, Jonn Axsen, and John W Polak. Modelling the influence of peers' attitudes on choice behaviour: Theory and empirical application on electric vehicle preferences. Transportation Research Part A: Policy and Practice, 140:278–298, 2020.
- [49] Katrin Merfeld, Mark-Philipp Wilhelms, and Sven Henkel. Being driven autonomously a qualitative study to elicit consumers' overarching motivational structures. <u>Transportation Research Part C:</u> Emerging Technologies, 107:229–247, 2019.

- [50] Vincent-Wayne Mitchell. Consumer perceived risk: conceptualisations and models. <u>European</u> Journal of Marketing, 33(1/2):163–195, Jan 1999.
- [51] Moataz Mohamed, Christopher D. Higgins, Mark Ferguson, and Weeberb J. Réquia. The influence of vehicle body type in shaping behavioural intention to acquire electric vehicles: A multi-group structural equation approach. Transportation Research Part A: Policy and Practice, 116:54–72, 2018.
- [52] Eric Molin, Kingsley Adjenughwure, Menno de Bruyn, Oded Cats, and Pim Warffemius. Does conducting activities while traveling reduce the value of time? evidence from a within-subjects choice experiment. Transportation Research Part A: Policy and Practice, 132:18–29, 2020.
- [53] Valentina Montoya-Robledo and Germán Escovar-Álvarez. Domestic workers' commutes in bogotá: Transportation, gender and social exclusion. <u>Transportation Research Part A: Policy and Practice</u>, 139:400–411, 2020.
- [54] Taka Morikawa, Moshe Ben-Akiva, and Daniel McFadden. Discrete choice models incorporating revealed preferences and psychometric data. Advances in Econometrics, 16:29–55, Jan 2002.
- [55] Yutaka Motoaki and Ricardo A. Daziano. A hybrid-choice latent-class model for the analysis of the effects of weather on cycling demand. <u>Transportation Research Part A: Policy and Practice</u>, 75:217– 230, 2015.
- [56] James A Mourey, Jenny G Olson, and Carolyn Yoon. Products as Pals: Engaging with Anthropomorphic Products Mitigates the Effects of Social Exclusion. <u>Journal of Consumer Research</u>, 44(2):414–431, 01 2017.
- [57] Pamela Murray-Tuite, Kris Wernstedt, and Weihao Yin. Behavioral shifts after a fatal rapid transit accident: A multinomial logit model. <u>Transportation Research Part F: Traffic Psychology and Behaviour</u>, 24:218–230, 2014.
- [58] Cornelia Măirean. Fear and avoidance of driving among drivers involved in a road traffic crash. the role of traumatic fear and driving cognitions. <u>Transportation Research Part F: Traffic Psychology and</u> Behaviour, 74:322–329, 2020.
- [59] Gopindra S. Nair and Chandra R. Bhat. Sharing the road with autonomous vehicles: Perceived safety and regulatory preferences. <u>Transportation Research Part C: Emerging Technologies</u>, 122:102885, jan 2021.
- [60] Zhipeng Niu, Xiaowei Hu, Shouming Qi, Haihua Yang, Siqing Wang, and Shi An. Determinants to parking mode alternatives: A model integrating technology acceptance model and satisfaction–loyalty model. Transportation Research Part A: Policy and Practice, 152:216–234, 2021.
- [61] Carol S. North, Aya Kawasaki, Edward L. Spitznagel, and Barry A. Hong. The course of ptsd, major depression, substance abuse, and somatization after a natural disaster. <u>The Journal of Nervous and</u> Mental Disease, 192(12), 2004.

- [62] Thomas A. Novack, Don Labbe, Miranda Grote, Nichole Carlson, Mark Sherer, Juan Carlos Arango-Lasprilla, Tamara Bushnik, David Cifu, Janet M. Powell, David Ripley, and Ronald T. Seel. Return to driving within 5 years of moderate–severe traumatic brain injury. <u>Brain Injury</u>, 24(3):464–471, 2010. PMID: 20184403.
- [63] Raymond W. Novaco, Daniel Stokols, Joan Campbell, and Jeannette Stokols. Transportation, stress, and community psychology. American Journal of Community Psychology, 7(4):361–380, Aug 1979.
- [64] Ashley Nunes, Bryan Reimer, and Joseph F. Coughlin. People must retain control of autonomous vehicles. Nature, 556:169–171, 2018.
- [65] Takanori Okada, Tetsuya Tamaki, and Shunsuke Managi. Effect of environmental awareness on purchase intention and satisfaction pertaining to electric vehicles in japan. <u>Transportation Research Part</u> D: Transport and Environment, 67:503–513, 2019.
- [66] P. Marijn Poortvliet and Anne Marike Lokhorst. The key role of experiential uncertainty when dealing with risks: Its relationships with demand for regulation and institutional trust. <u>Risk Analysis</u>, 36(8):1615–1629, 2016.
- [67] Dimitris Potoglou, Colin Whittle, Ioannis Tsouros, and Lorraine Whitmarsh. Consumer intentions for alternative fuelled and autonomous vehicles: A segmentation analysis across six countries. Transportation Research Part D: Transport and Environment, 79:102243, 2020.
- [68] Baiba Pudāne, Michał Rataj, Eric J.E. Molin, Niek Mouter, Sander van Cranenburgh, and Caspar G. Chorus. How will automated vehicles shape users' daily activities? insights from focus groups with commuters in the netherlands. <u>Transportation Research Part D: Transport and Environment</u>, 71:222– 235, 2019. The roles of users in low-carbon transport innovations: Electrified, automated, and shared mobility.
- [69] Weina Qu, Jing Xu, Yan Ge, Xianghong Sun, and Kan Zhang. Development and validation of a questionnaire to assess public receptivity toward autonomous vehicles and its relation with the traffic safety climate in china. Accident Analysis & Prevention, 128:78–86, jul 2019.
- [70] Neil Quarles, Kara M. Kockelman, and Jooyong Lee. America's fleet evolution in an automated future. Research in Transportation Economics, page 101107, 2021.
- [71] Alireza Rahimi, Ghazaleh Azimi, and Xia Jin. Examining human attitudes toward shared mobility options and autonomous vehicles. <u>Transportation Research Part F: Traffic Psychology and Behaviour</u>, 72:133–154, 2020.
- [72] Sebastián Raveau, Ricardo Álvarez Daziano, María Francisca Yáñez, Denis Bolduc, and Juan de Dios Ortúzar. Sequential and simultaneous estimation of hybrid discrete choice models: Some new findings. Transportation Research Record, 2156(1):131–139, 2010.

- [73] Zeinab Rezvani, Johan Jansson, and Jan Bodin. Advances in consumer electric vehicle adoption research: A review and research agenda. <u>Transportation Research Part D: Transport and Environment</u>, 34:122–136, 2015.
- [74] David Rojas-Rueda, Mark Nieuwenhuijsen, and Haneen Khreis. 1859 autonomous vehicles and public health: Literature review. Journal of Transport & Health, 5:S13, 2017.
- [75] Anthony J. Rosellini, Francisca Dussaillant, José R. Zubizarreta, Ronald C. Kessler, and Sherri Rose. Predicting posttraumatic stress disorder following a natural disaster. <u>Journal of Psychiatric Research</u>, 96:15–22, 2018.
- [76] David M. Sanbonmatsu, David L. Strayer, Zhenghui Yu, Francesco Biondi, and Joel M. Cooper. Cognitive underpinnings of beliefs and confidence in beliefs about fully automated vehicles. Transportation Research Part F: Traffic Psychology and Behaviour, 55:114–122, 2018.
- [77] Thomas W. Sanchez. Poverty, policy, and public transportation. <u>Transportation Research Part A:</u> Policy and Practice, 42(5):833–841, 2008.
- [78] E. Sardianou and P. Genoudi. Which factors affect the willingness of consumers to adopt renewable energies? Renewable Energy, 57:1–4, 2013.
- [79] Azim Shariff, Jean-François Bonnefon, and Iyad Rahwan. Psychological roadblocks to the adoption of self-driving vehicles. Nature Human Behaviour, 1(10):694–696, Oct 2017.
- [80] Ishant Sharma and Sabyasachee Mishra. Modeling consumers' likelihood to adopt autonomous vehicles based on their peer network. <u>Transportation Research Part D: Transport and Environment</u>, 87:102509, 2020.
- [81] Kong Joo Shin, Naoto Tada, and Shunsuke Managi. Consumer demand for fully automated driving technology. Economic Analysis and Policy, 61:16–28, 2019. Special issue on: Future of transport.
- [82] Frida Skarin, Lars E. Olsson, Margareta Friman, and Erik Wästlund. Importance of motives, selfefficacy, social support and satisfaction with travel for behavior change during travel intervention programs. Transportation Research Part F: Traffic Psychology and Behaviour, 62:451–458, 2019.
- [83] David Souder, Akbar Zaheer, Harry Sapienza, and Rebecca Ranucci. How family influence, socioemotional wealth, and competitive conditions shape new technology adoption. <u>Strategic Management</u> Journal, 38(9):1774–1790, 2017.
- [84] Araz Taeihagh and Hazel Si Min Lim. Governing autonomous vehicles: emerging responses for safety, liability, privacy, cybersecurity, and industry risks. Transport Reviews, 39(1):103–128, 2019.
- [85] Reza Vosooghi, Jakob Puchinger, Marija Jankovic, and Anthony Vouillon. Shared autonomous vehicle simulation and service design. <u>Transportation Research Part C: Emerging Technologies</u>, 107:15–33, oct 2019.

- [86] Gisela Wachinger, Ortwin Renn, Chloe Begg, and Christian Kuhlicke. The risk perception paradox—implications for governance and communication of natural hazards. <u>Risk Analysis</u>, 33(6):1049– 1065, 2013.
- [87] Zia Wadud, Don MacKenzie, and Paul Leiby. Help or hindrance? the travel, energy and carbon impacts of highly automated vehicles. <u>Transportation Research Part A: Policy and Practice</u>, 86:1–18, 2016.
- [88] Joan Walker and Moshe Ben-Akiva. Generalized random utility model. <u>Mathematical Social Sciences</u>, 43(3):303–343, 2002. Random Utility Theory and Probabilistic measurement theory.
- [89] Sicheng Wang, Zhiqiu Jiang, Robert B. Noland, and Andrew S. Mondschein. Attitudes towards privately-owned and shared autonomous vehicles. <u>Transportation Research Part F: Traffic</u> Psychology and Behaviour, 72:297–306, 2020.
- [90] Mark Warschauer. <u>Technology and Social Inclusion: Rethinking the Digital Divide</u>. The MIT Press, 09 2004.
- [91] Adam Waytz, Joy Heafner, and Nicholas Epley. The mind in the machine: Anthropomorphism increases trust in an autonomous vehicle. Journal of Experimental Social Psychology, 52:113–117, 2014.
- [92] Jingwen Wu, Hua Liao, Jin-Wei Wang, and Tianqi Chen. The role of environmental concern in the public acceptance of autonomous electric vehicles: A survey from china. <u>Transportation Research</u> Part F: Traffic Psychology and Behaviour, 60:37–46, 2019.
- [93] Zhigang Xu, Kaifan Zhang, Haigen Min, Zhen Wang, Xiangmo Zhao, and Peng Liu. What drives people to accept automated vehicles? findings from a field experiment. <u>Transportation Research Part C:</u> Emerging Technologies, 95:320–334, 2018.
- [94] Tingru Zhang, Da Tao, Xingda Qu, Xiaoyan Zhang, Rui Lin, and Wei Zhang. The roles of initial trust and perceived risk in public's acceptance of automated vehicles. <u>Transportation Research Part C:</u> Emerging Technologies, 98:207–220, 2019.
- [95] Tingru Zhang, Da Tao, Xingda Qu, Xiaoyan Zhang, Jihong Zeng, Haoyu Zhu, and Han Zhu. Automated vehicle acceptance in china: Social influence and initial trust are key determinants. <u>Transportation</u> Research Part C: Emerging Technologies, 112:220–233, 2020.
- [96] Fan Zhou, Zuduo Zheng, Jake Whitehead, Simon Washington, Robert K. Perrons, and Lionel Page. Preference heterogeneity in mode choice for car-sharing and shared automated vehicles. Transportation Research Part A: Policy and Practice, 132:633–650, 2020.
- [97] Jie Zhu and Ivana Tasic. Safety analysis of freeway on-ramp merging with the presence of autonomous vehicles. Accident Analysis & Prevention, 152:105966, mar 2021.
- [98] Álvaro Fernández-Heredia, Andrés Monzón, and Sergio Jara-Díaz. Understanding cyclists' perceptions, keys for a successful bicycle promotion. <u>Transportation Research Part A: Policy and Practice</u>, 63:1–11, 2014.

# Appendix

Table A1 presents the distribution of the socioeconomic variables in our sample and government statistics. Table A2 shows the average score (when we ask respondents to re-

		(n = 100, 803)	Government Statistics (70)
Gender	Female	41	51.3
	Male	59	48.7
Education level	Junior high school or less	2.1	9.5
	High school	26.9	42.3
	Some college	22.6	15.6
	Bachelor / Master / Doctor	45.9	23.9
	Other	1.9	8.6
Age	18–19	0.2	2.3
0	20–29	5.4	11.7
	30–39	18.1	13.3
	40–49	31.9	17.2
	50-64	25.8	22.1
	Over 65	10.7	33.4
Household income	<2 million JPY	7.8	18.3
	2–3 million JPY	8.9	17.2
	3–4 million JPY	11.9	15.3
	4–5 million JPY	12.3	12.2
	5–6 million JPY	11.9	9
	6–7 million JPY	9.6	6.9
	7–8 million JPY	9.1	5.8
	8–9 million JPY	6.9	4.1
	9–10 million JPY	6.7	3.4
	10–15 million JPY	10.5	6
	15–20 million JPY	2.7	1.1
	> 20 million JPY	1.7	0.7
	Don't know / don't want to answer	-	_
Region	Hokkaido	4.6	4.2
0	Tohoku	5.9	6.9
	Kanto	38.2	34.4
	Chubu	16.6	16.8
	Kinki	20.1	17.7
	Chugoku	5.1	5.8
	Shikoku	2.5	2.9
	Kyushu/Okinawa	7.1	11.3
Household size	1	15.6	34.5
	2	30.1	27.9
	3	23.6	17.6
	4 and above	30.1	20

Table A1: Socioeconomic Distribution of the Respondents and the Japanese Population

Sources: MIC (2017, 2019a, 2019b)

spond on 1-5 point scales) and proportion of respondents' evaluations of benefits and concerns regarding FAVs (for multiple choice questions). We calculate the proportions as follows: The number of people who choose the option/sample size (N=100,803).

Table A3 shows the results of the measurement equation, which describes the effects of the latent constructs on each of indicator variables. Standardized coefficients are shown, and all coefficients are positive and statistically significant at 0.1%. We only

#### Table A2: Proportion and Mean Value of Respondents' Evaluations for Each Latent Con-

struct.

Latent Category	All	Group 1	Group 2	Group 3
Latent Category 1: "Natural Disaster", 1-5 Point Scale		Mean	Value	
<b>ND1.</b> Do you think your life will be in danger due to a large-scale natural disaster?	2.31	2.32	2.3	2.13
ND2 Do you think your property (household household goods automobiles etc.)	2 11	2.13	2.01	2
will be demond by a large costs network (industrial disector)	2.11	2.15	2.01	2
will be damaged by a large-scale natural disaster?				
<b>ND3.</b> Do you think a large-scale natural disaster will	2.51	2.53	2.53	2.34
isolate you from your surroundings?				
<b>ND4.</b> Do you think you will need to evacuate from your home	2.73	2.74	2.72	2.64
during a disaster to save your life?				
Latent Category 2: "Social Support" 1-5 Doint Scale		Mean	Value	
Latent Gategory 2. Soliai Support, 1-5 rom to ale	2.20	2.25	2 41	2.40
SP1. 1 expect physical/mental support from the	2.30	2.35	2.41	2.49
government, local authorities and public institutions.				
<b>SP2.</b> I expect physical/mental support from family members and friends.	2.47	2.45	2.54	2.56
<b>SP3.</b> I expect physical/mental support from local	2.37	2.36	2.41	2.46
volunteers and members of local communities.				
Latent Category 3: "Merit", Multiple Choice		Evalua	tion (%)	
MB1. People can drive without a license	12 31%	12 08%	13 74%	12 75%
<b>MR2</b> Burdens on driving would be decreased	36 / 3%	36 12%	10.85%	32 9/1%
<b>MD2</b> . Dildren on move the vehicle without a guardian	2 0 407	2 0 4 07	40.0370	2 0 007
WRS. Children can move the venicle without a gladuan.	5.94%	3.04%	4.00%	3.90%
<b>MR4.</b> Able to do other work while driving. (Multitask)	27.96%	27.63%	32.08%	24.91%
<b>MR5.</b> Able to avoid responsibility for traffic accidents.	12.01%	11.84%	13.00%	12.50%
		т I	. (07)	
Latent Category 4: "Fear", Multiple Choice		Evalua	tion (%)	
<b>FE1.</b> There is a possibility that children will be able to move vehicles on their own.	33.74%	36.02%	39.09%	35.52%
<b>FE2.</b> There is a possibility that the software is hacked. (Cyber security)	65.13%	33.74%	37.85%	31.26%
<b>FE3.</b> A malfunction may cause accidents.	80.23%	53.04%	58.76%	45.29%
<b>FE4.</b> It is unclear who is responsible for an accident due to FAV technology.	76.63%	48.37%	53.04%	43.48%
<b>FES</b> Traffic volume and congestion might increase because those without a license can drive	52 98%	25.01%	26.91%	24 11%
<b>FEG</b> A malfunction may load ma to the urong doctination	51 20%	23.0170	20.3170	29.120%
<b>FEG.</b> A manufaction may lead me to the wrong destination.	31.20%	23.02%	20.10%	25.15%
Latent Category 5: "Pollution" 1-5 Point Scale		Mean	Value	
EDI Degrading is important	2 4 2	2 41	2 50	2 4 2
<b>EPD</b> . Recycling is important.	3.45	3.41	5.50	3.45
<b>EP2.</b> Cycle utilization rate, the percentage of the total amount of reusable and	3.34	3.31	3.48	3.35
recycled materials to be injected into society, is important for preventing pollution.				
<b>EP3.</b> I think water quality should be improved.	3.37	3.35	3.54	3.37
<b>EP4.</b> Alleviating particulate matter (PM) 2.5. pollution is critical for our society.	3.48	3.46	3.65	3.47
<b>EP5.</b> Resolving air pollution (particularly, photochemical smog) is important.	3.45	3.43	3.62	3.44
0 F				
Latent Category 6: "Nature", 1-5 Point Scale		Mean	Value	
EN1. Preserving endangered species is important.	2.94	2.92	3.04	3.01
<b>FN2</b> Preserving living animals (overall) is important	2.83	2.81	2 94	2.93
<b>EN9</b> The ratio of group area within 1 500 motors of a bausa is important	2.03	2.01	2.54	2.55
ENG. The fatto of green area within 1,300 meters of a house is important.	3.00	3.00	3.21	3.13
EN4. Green purchasing: when purchasing goods and services,	3.02	3	3.15	3.09
I consider the environmental impact before purchasing.				

present the measurement equation results of Model (4) because the coefficient values are almost unchanged across the five specifications including the robustness check. The results of the other specifications are available upon request. Note the need for caution when explaining the results in Tables 4, 5 and 6. While we observe generally positive correlation between the latent variables and indicator variables, we find exceptional evidence for ND. Such a difference suggests that if the respondent has high score on ND, s/he is not afraid of the risk of natural disasters. Therefore, a low score on ND can be regarded a respondent being afraid of natural disasters. On the other hand, for example, a high score on SP implies that the respondent is likely to expect social support.

	-	-	*
SP			
SP1	0.501	0 485	0.452
011	(0.004)	(0.005)	(0.006)
CDO	(0.004)	0.003)	0.000)
SPZ	0.574	0.563	0.560
	(0.004)	(0.005)	(0.006)
SP3	0.861	0.839	0.810
	(0.005)	(0.006)	(0.008)
ND			
ND1	-0.904	-0.871	-0.863
	(0.002)	(0.004)	(0.005)
ND2	0.002)	0.766	0.790
ND2	-0.800	-0.700	-0.780
NIDA	(0.002)	(0.004)	(0.005)
ND3	-0.672	-0.643	-0.645
	(0.003)	(0.004)	(0.005)
ND4	0.131	0.129	0.113
	(0.004)	(0.004)	(0.004)
FE			
FE1	0.524	0.513	0.518
	(0.002)	(0.005)	(0.005)
EEO	0.003)	0.005)	0.000
FE2	0.596	0.583	0.015
	(0.003)	(0.005)	(0.006)
FE3	0.610	0.611	0.625
	(0.003)	(0.005)	(0.006)
FE4	0.588	0.585	0.603
	(0.003)	(0.005)	(0.006)
FE5	0.511	0.502	0.525
	(0,002)	(0.005)	(0.006)
EEG	0.525	0.514	0.546
LC0	0.555	0.314	0.346
	(0.003)	(0.005)	(0.006)
MR			
MR1	0.413	0.395	0.405
	(0.004)	(0.005)	(0.006)
MR2	0.602	0.589	0.627
	(0.004)	(0.005)	(0.006)
MR3	0.371	0.345	0.370
	(0.005)	(0.005)	(0.006)
MR4	0.506	0.486	0.523
101114	0.300	0.400	0.323
MDF	(0.004)	(0.005)	(0.006)
MR5	0.421	0.404	0.410
	(0.004)	(0.005)	(0.006)
EP			
EP1	0.860	0.866	0.854
	(0.001)	(0.003)	(0.004)
EP2	0.849	0.840	0.838
-	(0.001)	(0.002)	(0.003)
ED3	0.800	0.89/	0.897
LIJ	(0.001)	0.034	0.007
ED4	(0.001)	(0.002)	(0.002)
EP4	0.937	0.938	0.923
	(0.001)	(0.001)	(0.002)
EP5	0.946	0.944	0.931
	(0.001)	(0.001)	(0.002)
EN			
EN1	0.884	0.877	0.877
	(0.001)	(0.002)	(0.003)
EN <sub>2</sub>	0.874	0.864	0.855
LINZ	(0.001)	0.004	0.000
ENIO	(0.001)	(0.002)	(0.003)
EN3	0.868	0.859	0.852
	(0.001)	(0.002)	(0.003)
EN4	0.828	0.820	0.817
	(0.001)	(0.003)	(0.004)

# Table A3: Coefficients of Measurement Equations Group 1 Group 2 Group 3 SP CD1 0.501 0.405 0.450

Note: All coefficients are

statistically significant at the 0.1% level.

#### Acknowledgements

Our special thanks go to Seilin Uhm, University College London, Institute of Education, Kyungwoong Koh, Faculty of Economics, Johns Hopkins University, and Akihiro Okuyama and Thierry Coulibaly, Urban Institute, Kyushu University and seminar participants at the University of Tokyo (Japan) and Kyushu University (Japan) for their helpful comments. This work was supported by JSPS KAKENHI Grant Number JP20H00648.

#### **Author Statement**

On behalf of all authors, the corresponding author states that there is no conflict of interest.

#### Contributions

Sunbin YOO: Conceptualization, Methodology, Formal analysis, Writing (Original Draft); Junya KUMAGAI: Methodology, Writing (Review and Editing); Yuta KAWABATA: Methodology; and Shunsuke MANAGI: Resources and Supervision.